\documentclass[12pt,a4paper]{article}
\usepackage[backref]{hyperref}
\usepackage{natbib}
%% Remarques : les ``packages'' définis dans le fichier aum2004.sty vous
%% permettent d'éditer directement les caractères français accentués.
%%
% Various bold symbols
\providecommand\bnabla{{\mathbf \nabla}} \providecommand\bcdot{{\mathbf \cdot}}
\begin{document}
%% Titre
\title{Wave-circulation-turbulence ... and remote sensing: some light reading}
%% Auteur
\author{Fabrice Ardhuin}
%% Adresse
%% Resume en francais : remplacer \cfmresumeexp{} par \cfmresume{}
%% pour supprimer dans le texte (gras, 12pt)
%\newpage
\textbf{A WISE bibliography on ocean waves}

Why a WISE bib?

Following the 10-year anniversary of the WAM Book (Komen et al. 1994), a white paper on research on
ocean wave forecasting and hindcasting was launched by Luigi Cavaleri as a collaborative effort of the
Waves In Shallow Environments group (WISE). In the
process, it became clear that the wide array of publications on wave-related
topics should be reviewed and old or not-so-old ideas reconsidered. Here is
thus an attempt to inventory these publications.

This effort may appear crazy and unnecessary to many, with the advent of specialized
search engines. However, these use indices such as the number of citations
which may not be able to highlight the really good stuff that nobody has read
nor cited. The following good old "manual" bibliography will hopefully be
helpful to colleagues that work on wave-related topics. The entries are sorted
by topic and then sorted by date of publication. A single entry should be
listed under different topics when appropriate.

 You can send your
own contributions with a bibliography in bibtex format to ardhuin(at)shom.fr. If you know of URLs where papers and reports
are openly available on the Internet, please send them so that the papers may be accessible at a single click from
this PDF document. The Oceanographical Society of Japan and the American Meteorological
Society are commended for their efforts to have all the "old" papers available to the general public at no cost.

How to use the WISE bib ?

Having grown over 2000 papers, with some items (such as bottom reflection and scattering) listing over 100, it seemed that
items should be split or more important papers be highlighted. While the former would lead to a larger fragmentation in
sub-specialties, the latter introduces a personal judgment and potential for endless rows with colleagues. (Why is my paper
less important than this one?). I will take that risk for now, and have thus chosen to \textbf{highlight in bold} a few landmark and
review papers that may be used as introduction to any sub-field. Suggestions are welcome.

This version was compiled on \textbf{\today}.


\section{General}
\begin{itemize}
\item  Relevant books and reviews \cite{Lamb1932}, \cite{Wehausen&Laitone1960},
\cite{Landau&Lifschitz1960}, \cite{Kinsman1965},
\cite{Jenkins&Watts1968}, \cite{Kitaigorodskii1970},
\cite{Whitham1974}, \cite{Roseau1976}, \cite{Phillips1977},
\cite{Lighthill1978}, \cite{LeBlondandMysak1978},
\cite{LeBlondandMysak1979}, \cite{Schlichting1979},
\cite{Priestley1981}, \cite{SWAMP}, \cite{Davidan&al.1985},
\cite{Goda1985}, \cite{Craik1985}, \cite{Stull1988},
\cite{Castillo1988}, \cite{Mei1989}, \cite{Rao&al.1990},
\cite{WAMBook}, \cite{Komar1998}, \cite{Ochi1998},
\cite{Dean&Dalrymple1991}, \cite{Nielsen1992},
\cite{Dingemans1997a}, \cite{Soulsby1997}, \cite{Young1999},
\cite{USACE2002}, \cite{Lavrenov2003}, \cite{Janssen2004},
\cite{WISE2007}

\item popular science texts \cite{Bascom1959}

\item Historical accounts on wave research \cite{Saint-Venant&Flamant1888}, \cite{Phillips1981a}, \cite{Ursell1999}, \cite{Craik2004}
\item a. Wave integral properties (energy, momentum, action, spin ...)  and variational principles
\cite{Starr1947a}, \cite{Starr1947b}, \cite{Platzman1947},
\cite{Phillips1961a}, \cite{Whitham1962}, \cite{Backus1962},
\cite{Whitham1965}, \cite{Penfield1966}, \cite{Hayes1970}, \cite{Jones1973},
\cite{Longuet-Higgins1975},
  \cite{Keady&Norbury1975}, \cite{Peregrine&Thomas1979}, \cite{Crapper1979}, \cite{Benzi&al.1979},
\cite{Longuet-Higgins1980b}, \cite{Benjamin&Olver1983},
\cite{Longuet-Higgins1983},
  \cite{Longuet-Higgins1984}, \cite{Yu&Wu1987}, \cite{Shields&Webster1988}, \cite{Benjamin1995},
\cite{Zakharov&Kusnetsov1997}, \cite{Jonsson1998}
  \item b. Wave kinematics (theory and verification)
   \cite{Poisson1818}, \cite{vonGerstner1809}, \cite{Airy1841}, \cite{Stokes1847}, \cite{Levi-Civita1925},
   \cite{Struik1926},
\cite{Miche1944b}, \cite{Friedrichs1948},
 \textbf{\cite{Ursell1953}}, \cite{Laitone1960}, \cite{Laitone1962},
\cite{Dean1965}, \cite{Thornton&Kraphol1974}, \cite{Schwartz1974},
\cite{Forristall&al.1978}, \cite{Rienecker&Fenton1981},
\cite{Lambrakos1981a}, \cite{Svendsen&Staub1981},
\textbf{\cite{Flick&al.1981}}, \cite{Battjes&vanHeteren1984},
\cite{Longuet-Higgins1986}, \cite{Longuet-Higgins1985},
  \cite{Ewing&al.1987}, \cite{Herbers&Guza1991},
\cite{Skjelbreia&al.1991}, \cite{Herbers&al.1992},
\cite{Herbers&Guza1992}, \cite{Kirby1998}, \cite{Baldock&Swan1996}
  \item c. Finite amplitude waves  \cite{Levi-Civita1925}
  \item d. non-Stokes waves \cite{vonGerstner1809}, \cite{Chen&Saffman1980},  \cite{Bryant1985}
  \item e. Dispersion relation
\cite{Laplace1776}, \cite{Stokes1847}, \cite{Rayleigh1877},
\cite{Sarpkaya1954}, \cite{Longuet-Higgins&Phillips1962},
\cite{Kutzbach&Bryson1963}, \cite{Whitham1967b},
\cite{Willebrand1973}, \cite{Willebrand1975},
  \cite{Huang&Tung1976},\cite{Weber&Barrick1977}, \cite{Barrick&Weber1977},
   \cite{Huang&Tung1977}, \cite{Masuda&al.1979}, \cite{Mitsuyasu&al.1979},
\cite{Plant&Wright1979}, \cite{Yefimov&Soloviev1979},
\cite{Ramamonjiarisoa&Mollo-Christensen1979},
\cite{Plant&Wright1980}, \cite{Komen1980}, \cite{Phillips1981b},
\cite{Herbers&al.2002}, \cite{Ehrenmark2005},
\cite{Madsen&Furham2006}
  \item f. Wave caustics \cite{Chao1971}, \cite{Chao&Pierson1972}, \cite{Pierson1972}, \cite{Brown2001}
  \item g. Mass and momentum of waves and currents and their interactions (general) \cite{Ursell1950},
  \cite{Whitham1962}, \cite{Backus1962}, \cite{Eckart1963}, \cite{Longuet-Higgins1969a},
  \cite{Bretherton1969}, \cite{Bretherton1969b},
  \cite{Hasselmann1970}, \cite{Pollard1970},
  \cite{Hasselmann1971}, \cite{Toba1972}, \cite{Peierls1976}, \cite{Phillips1977}, \cite{Madsen1977}, \cite{Andrews&McIntyre1976},
\cite{Andrews&McIntyre1976b}, \cite{Longuet-Higgins1977},
\cite{Andrews&McIntyre1978a},
  \cite{Andrews&McIntyre1978b}, \cite{Andrews&McIntyre1978b}, \cite{McIntyre1981}, \cite{Kenyon1983}, \cite{McIntyre1988},
  \cite{McIntyre&Norton1990}, \cite{Hasselmann1991}, \cite{Kirby&Lee1993},
  \cite{Kudryavtsev1994}, \cite{Holm1996}, \cite{Gjaja&Holm1996}, \cite{Yih1997}, \cite{Donelan1998}, \cite{Groeneweg&Klopman1998},
  \cite{Buhler&McIntyre1998}, \cite{Groeneweg1999},
  \cite{McWilliams&Restrepo1999},
   \cite{Lentz&al.1999}, \cite{Holm1999}, \cite{Buhler2000}, \cite{Phillips2001}, \cite{Perrie&al.2002}, \cite{Holm2002},
   \cite{Buhler&McIntyre2003}, \cite{Mellor2003},
   \cite{Ardhuin&al.2004a}, \cite{Jenkins&Ardhuin2004}, \cite{Jenkins2004},
   \cite{McWilliams&al.2004}, \cite{Wingate2004}, \cite{Kenyon2004}, \cite{Ardhuin2005},
\cite{Mellor2005}, \cite{Brostrom2005}, \cite{McIntyre&Mobbs2005},
\cite{Kenyon&Sheres2006}, \cite{Ardhuin2006b},
\cite{Ardhuin&al.2008}
  \item h. Wave transport equations \cite{Tatarski1961}, \cite{Andrews&McIntyre1978b},
\cite{Ryzhik&al.1996}
  \item i. Practical calculation of wave properties \cite{Hunt1979}, \cite{Wu&Thornton1986}, \cite{McKee1988}, \cite{Fenton&McKee1990}
  \item j. Well-posedness of the wave equations \cite{Lannes2005}
\end{itemize}
\section{Interaction of waves with the atmosphere}
\begin{itemize}
\item a. Relevant turbulence theory \cite{Taylor1938}, \cite{Townsend1950}, \cite{Batchelor&Proudman1954},
\cite{Phillips1955}, \cite{Frenkiel1962}, \cite{Wandel&Kofoed-Hansen1962},
\cite{Obukhov1962}, \cite{Lumley&Panofsky1964}, \cite{Townsend1970},
\cite{Hussain&Reynolds1970}, \cite{Bradshaw&Wong1972}, \cite{Davis1972},
\cite{Hunt1973}, \cite{Hunt&Graham1978}, \cite{Zilker&Hanratty1979},
\cite{Hino&al.1983}, \cite{Carruthers&Hunt1986}, \cite{Carruthers&Hunt1986},
\cite{Zeman&Jensen1987}, \cite{Hunt&Carruthers1990}, \cite{Durbin1993},
\cite{Mann1994}, \cite{Aronson&al.1997}, \cite{Meneveau&Katz2000}

\item b. Air-sea interactions: atmospheric boundary layer theory and wind profiles
\cite{Munk1947}, \cite{Monin1962}, \cite{Zilitinkevitch&Chalikov1968}, \cite{Dyer&Hicks1970},
\cite{Businger&al.1971}, \cite{Hsu&al.1994}, \cite{Lin&al.1996}, \cite{Miller&al.1999}, \cite{Zilitinkevitch&al.2001},
\cite{Sjoblom&Smedman2002}, \cite{Hara&Belcher2004}

\item c. Air-sea interactions: wind stress \cite{Charnock1955},
\cite{Stewart1961}, \cite{Toba1972}, \cite{Toba1973},
\cite{Banner&Melville1976}, \cite{Garratt1977},
\cite{Melville1977}, \cite{Donelan1979}, \cite{SSmith1980},
\cite{Large&Pond1981}, \cite{Donelan1982}, \cite{Wu1982},
\cite{Csanady1985}, \cite{Mitsuyasu1985}, \cite{Pacanowski1987},
\cite{Wu1988}, \cite{Gernaert1989}, \cite{Toba&al.1990},
\cite{Toba&Ebuchi1991}, \cite{Janssen1992}, \cite{SSmith&al.1992},
\cite{Ly1993}, \cite{Donelan&al.1993}, \cite{Gemmrich&al.1994},
\cite{Yelland&Taylor1996}, \cite{Anctil&Donelan1996},
\cite{Mahrt&al.1996}, \cite{Kitaigorodskii1997},
\cite{Donelan&al.1997}, \cite{Janssen1997},
\cite{Yelland&al.1998}, \cite{Rieder&Smith1998},
\cite{Donelan1998}, \cite{Banner&Peirson1998},
\cite{Hristov&al.1998}, \cite{Janssen1999},
\cite{Drennan&al.1999}, \cite{Drennan&al.1999b},
\cite{Smedman&al.1999}, \cite{Banner&al.1999},
\cite{Kudryavtsev&al.1999}, \cite{Chen&al.2001},
\cite{Taylor&Yelland2001} \cite{Grachev&Fairall2001},
\cite{Kudryavtsev&Makin2001}, \cite{Oost&al.2002},
\cite{Makin&Kudryavtsev.2002}, \cite{Janssen&al.2002},
\cite{Makin&Kudryavtsev.2002}, \cite{Lin&al.2002},
\cite{Drennan&al.2003}, \cite{Emanuel2003}, \cite{Makin&Stam2003},
\cite{Makin2003}, \cite{Mahrt&al.2003}, \cite{Smedman&al.2003},
\cite{Grachev&al.2003}, \cite{Fairall&al.2003},
\cite{Kudryavtsev&Makin2004}, \cite{Chelton&al.2004},
\cite{Andreas2004}, \cite{Lange&al.2004}, \cite{Guan&Xie2004},
\cite{Moon&al.2004}, \cite{vandenBrink&al.2004}, \cite{Hwang2005},
\cite{Papadimitrakis2005b}, \cite{Drennan&al.2005},
\cite{Adamson&al.2006}

\item c. Air-sea interactions: wind stress at high winds
\cite{Powell&al.2003}, \cite{Donelan&al.2004},
\cite{Bye&Jenkins2006}, \cite{Jarosz&al.2007},
\cite{Kudryavtsev2006}


\item d. Air flow separation above waves \cite{Banner&Melville1976}, \cite{Longuet-Higgins1990}, \cite{Giovanangeli&al.1999}

\item e. Waves and rain \cite{LeMehaute&Khangaonkar1990}, \cite{Poon&al.1992}

\item f. wind-wave generation and attenuation theories \cite{Jeffreys1925}, \cite{Jeffreys1926}, \cite{Phillips1957},
\cite{Miles1957}, \cite{Conte&Miles1959}, \cite{Benjamin1959},
\cite{Miles1959}, \cite{Miles1960}, \cite{Lighthill1962},
\cite{Miles1965}, \cite{Craik1966}, \cite{Hasselmann1967},
\cite{Stewart1967}, \cite{Longuet-Higgins1969},
\cite{Longuet-Higgins1969b}, \cite{Kendall1970},
\cite{Townsend1972}, \cite{Fabrikant1976}, \cite{Chalikov1976},
\cite{Valenzuela1976}, \cite{Dore1978}, \cite{Hasselmann1978},
\cite{Kawai1979}, \cite{Krasitskii1980}, \cite{Townsend1980},
\cite{Blennerhassett1980}, \cite{West&Seshadri1981},
\cite{Janssen1982}, \cite{Gelci&al.1985},
\cite{Nikolayeva&Tsimring1986}, \cite{Jacobs1987},
\cite{Janssen1989}, \cite{Beljaars&Taylor1989},
\cite{Weber&Forland1990}, \cite{Hara&Mei1991},
\cite{Belcher&Hunt1993}, \cite{Miles1993}, \cite{Chalikov1993},
\cite{Belcher&al.1994}, \cite{Miles1996a}, \cite{vanDuin1996a},
\cite{vanDuin1996b}, \cite{Miles1996b},
\cite{Walmsley&Taylor1996}, \cite{Miles&Ierley1998},
\cite{Cohen&Belcher1999}, \cite{Belcher1999},
\textbf{\cite{Miles1999}}, \cite{Sajjadi&Wakefield999},
\cite{Ierley&Miles2001}, \cite{Hara&Belcher2002},
\cite{Vanden-Broeck2002}, \cite{Abdalla&Cavaleri2002},
\cite{Abdalla&Bidlot2002}, \cite{McIntyre2003}, \cite{Janssen2004}

\item g. wind stress modulation by long waves \cite{Okuda&al.1977}, \cite{Garrett&Smith1976},
\cite{Sun&Wu1984}, \cite{Smith1986}, \cite{Jessup&Hesany1996},
\cite{Peirson&Banner2003}

\item h. numerical modeling of wind-wave coupling \cite{Gent&Taylor1976},
\cite{Chalikov1978}, \cite{Ichikawa1978},
\cite{AlZanaidi&Hui1984}, \cite{Chalikov1986}, \cite{Ly1991},
\cite{Chalikov&Makin1991}, \cite{Jenkins1992},
\cite{Chalikov1993}, \cite{Burgers&Makin1993},
 \cite{Chalikov&Belevich1993}, \cite{Harris&al.1996}, \cite{Mastenbroek1996}, \cite{Mastenbroek&al.1996}, \cite{Zou1998}, \cite{Makin&Kudryavtsev1999},
 \cite{Sullivan&al.2000}, \cite{Meirink&Makin2000},
 \cite{Kudryavtsev&al.2001},
\cite{Meirink&al.2003}

\item i. Observations of wind-wave interaction \cite{Snyder&Cox1966},  \cite{Harris1966}, \cite{Yefimov1970}, \cite{Dobson1971},
\cite{Elliott1972a}, \cite{Elliott1972b}, \cite{Davidson&Franck1973},
\cite{Larson&Wright1975}, \cite{Hsiao&Shemdin1983}, \cite{Snyder&al.1981},
\cite{Mitsuyasu&Honda1982}, \cite{Plant1982}, \cite{Young&Sobey1985},
\cite{Giovanangeli&Memponteil1985}, \cite{Banner1990b},
\cite{Hasselmann&Bosenberg1991}, \cite{Mastenbroek&al.1996},
\cite{Hristov&al.1998}, \cite{Giovanangeli&al.1999}, \cite{Hristov&al.1999},
\cite{Donelan1999}, \cite{Hristov&al.2003}, \cite{Peirson&al.2003},
\cite{Donelan&al.2005a}, \cite{Donelan&al.2005b}, \cite{Feddersen&Veron2005},
\cite{Agnon&al.2005}, \cite{Donelan&al.2006}

\item j. Negative wind input \cite{Harris1966}, \cite{Grachev&Fairall2001}, \cite{Grachev&al.2003}, \cite{Kudryavtsev&Makin2004}

\item k. Wind input parameterization \cite{WAMDI1988}, \cite{Janssen1991}, \cite{Jenkins1993},
\cite{Janssen&al.1994}, \cite{Tolman&Chalikov1996},
\cite{Myrhaug&Slaattelid2003}, \cite{Janssen2004}

\item l. Air-sea interactions: sea state and heat fluxes \cite{Friehe&Schmitt1976}, \cite{Wu1992}, \cite{Mahrt&al.1998},
\cite{Makin1998}, \cite{Andreas&Monahan2000}

\item m. Spray \cite{Anguelova&al.1999}, \cite{Wu2000a}

\item n. Gas exchange \cite{Csanady1990}, \cite{Memery&al.2002}, \cite{Jahne&Haussecker1998}
\end{itemize}


\section{Wave-wave interactions}
\begin{itemize}
\item Short wave - long wave modulation
\cite{Longuet-Higgins&Stewart1960}, \cite{Phillips1963},
\cite{Hasselmann1971}, \cite{Keller&Wright1975},
\cite{Wright1976}, \cite{Garrett&Smith1976},
\cite{Alpers&Hasselmann1978}, \cite{Valenzuela&Wright1979},
\cite{Wu1979}, \cite{Longuet-Higgins1980c}, \cite{Pierson1980},
\cite{Monaldo&Kasevich1981b}, \cite{Bryant1982}, \cite{Monaldo&Kasevich1982},
\cite{Longuet-Higgins1985}, \cite{Smith1986},
\cite{Longuet-Higgins1987a}, \cite{Longuet-Higgins1987c},
\cite{Henyey&al.1988}, \cite{Naciri&Mei1992}
\cite{Cavaleri&Lionello1992}, \cite{Chu&al.1992},
\cite{Hara&Plant1994}, \cite{Kudryavtsev1994},
\cite{Jessup&Hesany1996}, \cite{Kudryavtsev&al.1997},
\cite{Chen&Belcher2000}, \cite{Elfouhaily&al.2001},
\cite{Elfouhaily&al.2001b}, \cite{Besse&Lannes2001},
\cite{Kudryavtsev&al.2003b}, \cite{Charnotskii&al.2002},
\cite{Hara&al.2003}, \cite{Gorshkov&al.2003},
\cite{Pan&Johnson2006}

\item Capillary waves interaction with gravity waves \cite{Longuet-Higgins1963}
\cite{Davies&Vose1965}, \cite{McGoldrick1965}, \cite{Fedorov&Melville1998}

\item 4 and 5 wave interactions theory \textbf{\cite{Phillips1960}}, \cite{Hasselmann1960}, \cite{Phillips1961a},
\cite{Phillips1961b}, \cite{Hasselmann1961},
\cite{Vedenov&al.1961}, \cite{Longuet-Higgins1962},
\cite{Armstrong&al.1962}, \textbf{\cite{Hasselmann1962}},
\cite{Benney1962}, \cite{Hasselmann1963a}, \cite{Bretherton1964},
\cite{McGoldrick1965}, \textbf{\cite{McGoldrick&al.1966}},
\cite{Hasselmann1966}, \cite{Benney&Saffman1966},
\cite{Phillips1967}, \cite{Zakharov1967},
\cite{Newell&Aucoin1971}, \cite{Zakharov&Shabat1972},
\cite{Phillips1974}, \cite{Herterich&Hasselmann1980},
\cite{Phillips1981a}, \cite{Yuen&Lake1982},
\cite{Hammack&Henderson1993}, \cite{Krasitskii1994},
\cite{Elfouhaily&al.2000}, \cite{Tanaka2001a}, \cite{Tanaka2001b},
\cite{Stiassnie2004}, \cite{Janssen2004},
\cite{Annenkov&Shrira2006}

\item horseshoe and other 2D wave patterns
\cite{Kusaba&Mitsuyasu1986}, \cite{Hammack&al.1989},
\cite{Hammack&al.1995}, \cite{Milewski&al.1996},
\cite{Hammack&al.2005}, \cite{Collard&Caulliez1999},
\cite{Furham&Madsen2006}, \cite{Henderson&al.2006}

\item four wave interactions numerical calculations and parameterization
\cite{Hasselmann1963b}, \cite{Longuet-Higgins1976},
\cite{Fox1976}, \cite{Webb1978}, \cite{Dungey&Hui1979},
\cite{Hasselmann&Hasselmann1985}, \cite{Hasselmann&al.1985b},
\cite{Tracy&Resio1982}, \cite{Zakharov&Zaslavskii1982},
\cite{Young&al.1987}, \cite{Resio&Perie1991},
\cite{Davidan&Lavrenov1991}, \cite{Zakharov1992},
\cite{Young&VanVledder1993}, \cite{Vledder&Holthuijsen1993},
\cite{Snyder&al.1993}, \cite{Masson1993},
\cite{Komatsu&Masuda1996}, \cite{Snyder&al.1998},
\cite{Balk&Zakharov1998}, \cite{Lin&Perrie1999},
\cite{Zakharov&Pushkarev1999}, \cite{Zaslavskii2000},
\cite{Vledder&al.2000}, \cite{Lin&Su2000}, \cite{Lavrenov2001},
\cite{Krasnopolsky&al.2001}, \cite{Polnikov2003},
\cite{Gorman2003}, \cite{Pushkarev&al.2004},
\cite{Tolman&al.2005}, \cite{Benoit2005}, \cite{vanVledder2006},
\cite{Resio&Perrie2007}

\item "wave" or "weak" turbulence \cite{Zakharov&al.2004},
\cite{Dyachenko&al.2004}

\item Wave instabilities \cite{Benjamin&Feir1967}, \cite{Whitham1967}, \cite{Feir1967},
\cite{Benjamin1967}, \cite{Longuet-Higgins&Cokelet1976},
\cite{Lake&al.1977}, \cite{Alber1978},
\cite{Longuet-Higgins1978b}, \cite{Longuet-Higgins&Cokelet1978},
\cite{McLean1982}, \cite{Tanaka1983}, \cite{Melville1983},
\cite{Tanaka1983}, \cite{Chereskin&MolloChristensen1985},
\cite{Tanaka1985}, \cite{Bliven&al.1986}, \cite{Li&al.19XX},
\cite{Tulin&Waseda1999}, \cite{vanDuin1999},
\cite{Brown&Jensen2001}, \cite{Janssen2004}, \cite{Chalikov2007}

\item other instabilities \cite{Larsen1979}, \cite{Bryant1989}

\item 3-wave interactions and nonlinear shoaling \cite{Fermi&al.1955},
\cite{McGoldrick1965}, \cite{Peregrine1967}, \cite{Lau&Barcilon1972},
\cite{Craik&Adam1978}, \cite{Guza&Thornton1980}, \cite{Miles1983},
\cite{Freilich&Guza1984}, \cite{Iusim&Stiassnie1985}, \cite{Elgar&Guza1985a},
\cite{Elgar&Guza1985b}, \cite{Elgar&Guza1986},
 \cite{Iusim&Stiassnie1985}, \cite{Freilich&al.1990}, \cite{Elgar&al.1990}, \cite{Abreu&al.1992}, \cite{Grue1992},
 \cite{Elgar&al.1993},
\cite{Beji&Battjes1993}, \cite{Beji&Battjes1994},
\cite{Eldeberky&Battjes1995}, \cite{Herbers&Burton1997},
\cite{Agnon&Sheremet1997}, \cite{Norheim&al.1998},
\cite{Young&Eldeberky1998}, \cite{Zaslavskii&Polnikov1998},
\cite{Becq1998}, \cite{Agnon&al.1999}, \cite{BecqGirard&al.1999},
\cite{Agnon&Sheremet2000}, \cite{Gobbi&al.2000},
\cite{Herbers&al.2003}, \cite{Benilov&al.2005},
\cite{Polnikov2005a}, \cite{Furham2004}, \cite{Janssen&al.2006},
\cite{Janssen2006}


\end{itemize}

\section{Nonlinear wave models}
\begin{itemize}

\item Nonlinear shallow water and others \cite{Green&Naghdi1976}, \cite{Didenkulova&al.2006}

\item Solutions to Euler's equations (irrotational flow)
\cite{Craig&Sulem1993}, \cite{Grilli&al.1997},
\cite{Fochesato&Dias2006}

\item Zakharov equation \cite{Annenkov&Shrira1999}, \cite{Rasmussen&Stiassnie1999},
\cite{Annenkov&Shrira2001}, \cite{Annenkov&Shrira2001b},
\cite{Willemsen2001}, \cite{Kit&Shemer2002}, \cite{Willemsen2002},
\cite{Janssen2004}, \cite{Janssen&Onorato2005},
\cite{Guyenne&Nichols2005}, \cite{Annenkov&Shrira2006}

\item Time-domain Boussinesq and Serre equations
\cite{Peregrine1967}, \cite{Nwogu1993}, \cite{Wei&al.1995},
\cite{Nadaoka&al.1997}, \cite{Gobbi&Kirby1999},
\cite{Bellotti&Brocchini2001}, \cite{Barthelemy2004},
\cite{Liu&Sun2005}, \cite{Musumeci&al.2005}, \cite{Cienfuegos2005}

\item Multi-layer Boussinesq
\cite{Lynett&Liu2005}

\item KdV and KP equations and solutions \cite{Boussinesq1872}, \cite{Korteweg&deVries1895},
\cite{Wiegel1959}, \cite{Johnson1973}, \cite{Green&Naghdi1976},
\cite{Svendsen&Hansen1978}, \cite{Anker&Freeman1978},
\cite{Fornberg&Whitham1978}, \cite{Freeman1980}, \cite{Miles1981},
\cite{Freeman&Nimmo1983}, \cite{Kirby&Vengayil1988},
\cite{Furzeland&al.1990}, \cite{Osborne&Segre1993},
\cite{Osborne1993}, \cite{Osborne1995}, \cite{Abdullaev&al.1995},
\cite{Osborne&al.1998}, \cite{Feng&Mitsui1998},
\cite{Feng&Kawahara2000a}, \cite{Feng&Kawahara2000b}

\item nonlinear Schr{\"o}dinger equation and solutions \cite{Hasimoto&Ono1972}, \cite{Yuen&Ferguson1978},
\cite{Chereskin&MolloChristensen1985}, \cite{Ablowitz&Villarroel1997},
\cite{Shagalov1998}, \cite{Dorren1999}, \cite{Stocker&Peregrine1999a},
\cite{Osborne&al.2003}, \cite{Onorato&al.2003}, \cite{Islas&Schober2005}

\item higher order spectral methods \cite{Fructus&al.2005},
\cite{Johannessen&Swan2003}

\end{itemize}

\section{Interaction of waves with the upper ocean}
\begin{itemize}
\item Waves and the Ekman layer \cite{Ekman1905}, \cite{Blackadar1962},
\cite{Madsen1977}, \cite{Jones&Kenney1977}, \cite{Madsen1978},
\cite{Yamagata1978}, \cite{Csanady&Shaw1980}, \cite{Weber1981},
\cite{Weber1983a}, \cite{Weber1983b}, \cite{Jenkins1986},
\cite{Jenkins1987}, \cite{Price&al.1987}, \cite{Bye1988},
\cite{Jenkins1989}, \cite{Coleman&al.1990},
\cite{Weber&Melsom1993a}, \cite{Weber&Melsom1993b},
\cite{Xu&Bowen1994}, \cite{Romanou&Weatherly1999},
\cite{Plueddemann&Weller1999}, \cite{Weber2003},
\cite{Lewis&Belcher2004}, \cite{Ardhuin&al.2004b},
\cite{Polton&al.2005}, \cite{Maximenko&Gouskina2005},
\cite{Bronstrom2005}, \cite{Polnikov&Tkalich2006},
\cite{Rascle&al.2006}, \cite{Rascle2007}

\item Wave effects on the global ocean \cite{McWilliams&Restrepo1999}, \cite{Qiao&al.2004},
\cite{Wang&Huang2004a}, \cite{Wang&Huang2004b}


 \item Surface drift \cite{Pierson1962}, \cite{LeMehaute1968}, \cite{Kenyon1969}, \cite{Alofs&Reisberg1972},
\cite{Shemdin1972},
\cite{Ianiello&Garvine1975}, \cite{Wright1976},
\cite{Dobroklonsky&Lesnikov1975},
 \cite{Wu1975}, \cite{Kondo1976}, \cite{Kraus1977}, \cite{Lange&Huhnerfuss1978}, \cite{Huang1979},
\cite{Longuet-Higgins1979}, \cite{Longuet-Higgins1979b}, \cite{Schooley1979}, \cite{Tsahalis1979},
\cite{Wu1983}, \cite{Churchill&Csanady1983}, \cite{Csanady1984},
 \cite{Longuet-Higgins1987b}, \cite{Richman&al.1987}, \cite{Melville&Rapp1988},
\cite{Elliott1991}, \cite{ValleLevinson&Swanson1991}, \cite{Santala1991},
 \cite{Santala&Terray1992},
 \cite{Youssef&Spaulding1993},
\cite{Gnanadesikan&Weller1995}, \cite{Melsom1996},
\cite{Kank&Lee1996}, \cite{Eames&McIntyre1999},
\cite{Spaulding1999}, \cite{Weber2001}, \cite{Bye&Wolff2001},
\cite{Annika&al.2001}, \cite{Restrepo&Leaf2002},
\cite{Thompson&al.2003}, \cite{Peirson&Banner2003},
\cite{Lewis&Belcher2004}, \cite{Ng2004},
\cite{Monismith&Fong2004}, \cite{Ardhuin&al.2004b},
\cite{Ogasawara&Yasuda2004}, \cite{Jenkins&Ward2005},
\cite{Smith2006b}, \cite{Rascle&al.2006}, \cite{Huang2006},
\cite{Monismith&al.2007}, \cite{Rascle2007}

 \item Viscous layer \cite{McLeish&Putland1975}, \cite{Okuda&al.1976}, \cite{Wu1984},
\cite{Peirson&Banner2003}

\item Waves on vertically sheared currents
\cite{Biesel1950}, \cite{Burns1953}, \cite{Sarpkaya1954},
\cite{Craik1968}, \cite{Yih1972}, \cite{Dalrymple1974},
\cite{Peregrine1976}, \cite{Jonsson&al.1978}, \cite{Thomas1981},
\cite{Kemp&Simons1982}, \cite{Kemp&Simons1983}, \cite{Ismail1984},
\cite{Srokosz1985}, \cite{Skop1987},
\cite{TelesdaSilva&Peregrine1988}, \cite{Kirby&Chen1989},
\cite{Jonsson1990}, \cite{Shrira1993}, \cite{White1999},
\cite{Swan&al.2001}, \cite{Miles2001a}, \cite{McWilliams&al.2004}

\item Waves in non-homogeneous media \cite{Whitham1967}, \cite{Garrett1967}, \cite{Lighthill1967}, \cite{Bretherton&Garrett1968},
 \cite{Willebrand1973}, \cite{Willebrand1975}, \cite{Voronovich1976}, \cite{Kravtsov&Orlov1980},
 \cite{Voronovich&Goncharov1982}, \cite{vanDuin1999},
\cite{Lavrenov2003}

\item Waves in random media \cite{Wigner1932}, \cite{Bourret1962}, \cite{Howe1971a},
\cite{Howe1971b}, \cite{Watson&West1975}, \cite{Rayevskiy1983},
\cite{Fabrikant&Raevsky1994}, \cite{Ryzhik&al.1996}, \cite{Gerard&al.1997},
\cite{Bal&al.1999}, \cite{Ardhuin&Herbers2002}, \cite{Ardhuin&Magne2007}

\item Waves and surges or tidal elevations \cite{Tancreto1958},
\cite{Booij1981}, \cite{Tolman1988}, \cite{Tolman1990a}, \cite{Tolman1990b},
\cite{Tolman1990c}, \cite{Tolman1991a}, \cite{Schneggenburger&al.2000}

\item Waves on horizontally varying currents \cite{Johnson1947}, \cite{Isaacs1948}, \cite{Hughes&Stewart1961},
\cite{Longuet-Higgins&Stewart1961}, \cite{Cavanie&al.1982},
\cite{McKee1974}, \cite{McKee1975}, \cite{Evans1975},
\cite{Peregrine&Smith1975}, \cite{Long&Huang1976},
\cite{McKee1978}, \cite{Mollo-Christensen1978},
\cite{Peregrine&Thomas1979}, \cite{Vincent1979},
\textbf{\cite{Stiassnie&Peregrine1979}}, \cite{Brevik&Aas1980},
\cite{Hayes1980}, \cite{Ismail1980}, \cite{Lambrakos1981b},
\cite{Ismail1981}, \cite{Christoffersen&Jonsson1981},
\cite{Booij1981}, \cite{McClain&al.1982}, \cite{Smith1983},
\cite{Liu1983}, \cite{Meadows&al.1983}, \cite{Ismail&Wiegel1983},
\cite{Rayevskiy1983}, \cite{Mei&Lo1984}, \cite{Kirby1984},
\cite{Gonzalez1984}, \cite{Phillips1984}, \cite{Mapp&al.1985},
\cite{Lavrenov1986}, \cite{McKee1986},
\cite{Gutshabash&Lavrenov1986}, \cite{McKee1987},
\cite{McKee&Tesoriero1987}, \cite{Kirby&al.1987},
\cite{Smith1987}, \cite{Irvine&Tilley1988},
\cite{Shyu&Phillips1990}, \cite{Liu&al.1990},
\cite{Sheres&Kenyon1990}, \cite{Holthuijsen&Tolman1990},
\cite{Hubbert&Wolf1991}, \cite{Tolman1991a},
\cite{Holthuijsen&Tolman1991}, \cite{Tolman1992c},
\cite{Verhagen&al.1992}, \cite{Gerber1993},
\cite{Trulsen&Mei1993}, \textbf{\cite{Fabrikant&Raevsky1994}},
\cite{McKee1994}, \cite{Kudryavtsev&al.1995}, \cite{McKee1996},
\cite{Masson1996}, \cite{White&Fornberg1998}, \cite{Shyu&al.1999},
\cite{Shyu&Tung1999}, \cite{Wolf&Prandle1999}, \cite{White1999},
\cite{Grodsky&al.2000}, \cite{Dysthe2001}, \cite{Pihl&al.2001},
\cite{Osuna2002}, \cite{Hisaki2002}, \cite{Bal&Chou2002},
\cite{McKee2003}, \cite{Lin&Perrie2003}, \cite{Lin&Chubb2003},
\cite{Belibassakis&Athanassoulis2004b}, \cite{Belibassakis2005},
\cite{McKee2006}, \cite{Hwang&al.2006}, \cite{Belibassakis2007}

\item Wave blocking \cite{Lai&al.1989}, \cite{Shyu&Phillips1990},
\cite{Ris&Holthuijsen1996}, \cite{Shyu&al.1999}, \cite{Chawla&Kirby2002},
\cite{Chawla&Kirby2005}, \cite{Suastika&Battjes2005}

\item Radiation and wave-induced stresses
\textbf{\cite{Longuet-Higgins&Stewart1964}}, \cite{Battjes1972},
\cite{Mei1973}, \cite{James1974} \cite{Andrews&McIntyre1978b},
\textbf{\cite{Stiassnie&Peregrine1979}}, \cite{Herbers&Guza1989},
\cite{deVriend&Kitou1990}, \cite{Smith1990}, \cite{Deigaard1993},
\textbf{\cite{Rivero&Arcilla1995}}, \cite{You1997},
\cite{Rivero&Arcilla1997}, \cite{Nobuoka&Mimura2002},
\cite{Nobuoka&Mimura2003}, \cite{Mellor2003}, \cite{Hughes2004},
\cite{Nobuoka&al.2004}, \cite{Xia&al.2004},
\textbf{\cite{McWilliams&al.2004}}, \cite{Ardhuin2005},
\cite{Smith2006a}, \cite{Zou&al.2006}, \cite{Ardhuin&Rascle2006},
\cite{Shi&al.2006}, \cite{Lane&al.2007}, \cite{Ardhuin&al.2008},
\cite{Newberger&Allen2007a}, \cite{Ardhuin&al.2008}

\item Mass transport and wave boundary layers (streaming)
\cite{Caligny1878}, \cite{Longuet-Higgins1953}, \cite{Russel&Osorio1958},
\cite{Longuet-Higgins1960}, \cite{Collins1963}, \cite{Johns1970},
\cite{Huang1970}, \cite{Unluata&Mei1970}, \cite{Johns1970}, \cite{Sleath1973},
\cite{Liu&Davis1977}, \cite{Johns1977}, \cite{Lamoure&Mei1977},
\cite{Yamagata1978}, \cite{Trowbridge&Madsen1984b}, \cite{Weber1985},
\cite{Sakakiyama&Bijker1989}, \cite{Iskandarani&Liu1991}, \cite{Mei&Chian1994},
\cite{Piedra-Cueva1995}, \cite{Vittori&Blondeaux1996},
\cite{Davies&Villaret1999}, \cite{Marin2004}

\item deep-water wave breaking: kinematics and statistics \cite{Stokes1880}, \cite{Miche1944d}, \cite{Donelan&al.1972},
\cite{Longuet-Higgins1973}, \cite{Longuet-Higgins&Turner1974},
\cite{Longuet-Higgins&Fox1977}, \cite{Cokelet1977},
\cite{Duncan1981}, \cite{Battjes&Sakai1981},
\cite{Longuet-Higgins1982b}, \cite{Duncan1983},
\cite{Greenhow1983}, \cite{Holthuijsen&Herbers1986},
\cite{Melville&Rapp1988}, \cite{Dommermuth&al.1988},
\cite{Tung&al.1989}, \cite{Hwang&al.1989}, \cite{Bonmarin1989},
\cite{Katsaros&Atakturk1992}, \cite{Cointe&Tulin1994},
\cite{Jenkins1994}, \cite{Ding&Farmer1994}, \cite{Skyner1996},
\cite{Melville1996}, \cite{Smith&al.1996},
\cite{Chanson&Jaw-Fang1997}, \cite{Peregrine1999},
\cite{Gemmrich&Farmer1999b} \cite{Hanson&Phillips1999},
\cite{Jenkins2000}, \cite{Meza&al.2000}, \cite{Banner&al.2000},
\cite{Andrew&al.2001}, \cite{Babanin&al.2001},
\cite{Banner&al.2002}, \cite{Melville&al.2002},
\cite{Steinbach&al.2002}, \cite{Liu&Babanin2004},
\cite{Papadimitrakis2005a}

\item deep-water wave breaking: instabilities, thresholds and breaking criteria
\cite{Miche1944a}, \textbf{\cite{Miche1944d}},
\cite{Banner&Phillips1974}, \cite{Phillips&Banner1974},
\cite{Wright1976}, \cite{Nath&Ramsey1976}, \cite{Melville1982},
\cite{Snyder&Kennedy1983}, \cite{Kennedy&Snyder1983},
\cite{Snyder&al.1983}, \cite{Koga1984}, \cite{Weissman&al.1984},
\cite{Longuet-Higgins1985}, \cite{Tanaka&al.1987},
\cite{Longuet-Higgins1988}, \cite{Longuet-Higgins1988},
\cite{Longuet-Higgins1990}, \cite{Banner&Peregrine1993},
\cite{Longuet-Higgins&Cleaver1994},
\cite{Longuet-Higgins&al.1994}, \cite{Schultz&al.1994},
\cite{Longuet-Higgins&Dommermuth1997}, \cite{Banner&Tian1998},
\cite{Song&Banner2002}, \cite{Stansell&MacFarlane2002},
\cite{Banner&al.2004}, \cite{Oh&al.2005}, \cite{Yao&Wu2005},
\cite{Banner&Peirson2007}

\item microscale breaking
\cite{Jessup&al.1997}, \cite{Jessup&Phadnis2005},
\cite{Siddiqui&Loewen2007}

\item breaking and energy dissipation \cite{Duncan1981}, \cite{Duncan1983}, \cite{Melville1994},
\cite{Kraan&al.1996}

\item Wave breaking and frequency downshift \cite{Tulin&Waseda1999}

\item wave breaking and bubbles \cite{Koga1982}, \cite{Hwang&al.1990}, \cite{Lamarre&Melville1991}, \cite{Hwang&al.1991}, \cite{Thorpe&al.1992},
\cite{Vagle&Farmer1998}, \cite{Phelps&Leighton1998},
\cite{Terrill&Melville2000}, \cite{Wu2000b},
\cite{Garrett&al.2000}, \cite{Chanson2002},
\cite{Chanson&al.2002b}, \cite{Deane&Stokes2002},
\cite{Graham&al.2004}, \cite{Graham2004}, \cite{Manasseh&al.2006}

\item breaking probability, whitecaps and breaking crests coverage \cite{Collins1970},
\cite{Wu1979b}, \cite{Monahan&OMuircheart1981},
\cite{Weissman&al.1984}, \cite{Farmer&Vagle1988}, \cite{Wu1988b},
\cite{Hanson&Phillips1999}, \cite{Zhao&Toba2001},
\cite{Melville&Matusov2002}, \cite{Reul&Chapron2003},
\cite{Stramska&Petelski2003}, \cite{Lafon&al.2004},
\cite{Anguelova&Webster2006}

\item Spectral signature of breaking waves
\cite{Nepf&al.1998}, \cite{Young&Babanin2006b}


\item deep-water wave breaking parameterizations \cite{Hasselmann1974}, \cite{Komen&al.1984}, \cite{Phillips1984},
\cite{Banner&Young1994}, \cite{Donelan&Yuan1994},
\cite{Tolman&Chalikov1996}, \cite{Alves2000},
\cite{Holthuijsen&al.2000}, \cite{Donelan2001},
\cite{Booij&Holthuijsen2002}, \cite{vanVledder&Hurdle2002},
\cite{Alves&Banner2003}, \cite{Rogers&al.2003},
\cite{Banner&al.2004}, \cite{Lefevre&al.2004},
\cite{Bidlot&al.2005}, \cite{vanderWesthuysen&al.2005},
\cite{Young&Babanin2006b}, \cite{Banner&Morison2006},
\cite{vanderWesthuysen&al.2007}, \cite{Bidlot&al.2007}

\item observations of surface mixing and theory \cite{Belyaev&al.1975},
\cite{Arsenyev&al.1975}, \cite{Richman&Garrett1977},
\cite{Price1979}, \cite{Dillon&al.1981}, \cite{Oakey&Elliott1982},
\cite{Thorpe1984a}, \cite{Thorpe1984b}, \cite{dAsaro1985},
\cite{Shay&Gregg1986}, \cite{Narimousa&al.1986},
\cite{Cavaleri&Zecchetto1987}, \cite{Soloviev&al.1988},
\cite{Delvigne&Sweeney1988}, \cite{Kudryavtsev&Soloviev1990},
\cite{Sukhorukov&Dmitriev1990}, \cite{Agrawal&al.1992},
\cite{Kitaigorodskii1994}, \cite{Anis&Moum1995},
\cite{dAsaro&al.1995}, \cite{Terray&al.1996}, \cite{Garrett1996},
\cite{Drennan&al.1996}, \cite{Burgers1997},
\cite{Kitaigorodskii1997}, \cite{Stacey&Pond1997},
\cite{Skyllingstad&al.1999}, \cite{Smith&al.1999},
\cite{Kocsis&al.1999}, \cite{Gemmrich&Farmer1999},
\cite{Stacey1999}, \cite{Gemmrich2000}, \cite{Terray&al.2000},
\cite{Kitaigorodskii2001}, \cite{Greenan&al.2001},
\cite{Thorpe&al.2003a}, \cite{Chukharev2003}, \cite{dAsaro2003},
\cite{Soloviev&Lukas2003}, \cite{Gemmrich&Farmer2004},
\cite{Tseng&dAsaro2004}, \cite{Zilintikevich&Esau2005},
\cite{Babanin&al.2005}, \cite{Thorpe&Osborn2005},
\cite{Babanin2006}, \cite{LeQuere&al.2007},
\cite{Sriver&Huber2007}

\item Parameterization of surface mixing in the ocean (see also www.gotm.net)
\cite{Mellor&Yamada1974}, \cite{Garwood1977},
\cite{Klein&Coantic1981}, \cite{Mellor&Yamada1982},
\cite{Andre&Lacarrere1985}, \cite{Ly1986}, \cite{Ly1990},
\cite{Gaspar&al.1990}, \cite{Large&al.1994},
\cite{Craig&Banner1994}, \cite{Lentz1995}, \cite{Li&al.1995},
\cite{Noh1996}, \cite{Craig1996}, \cite{Stacey&Pond1997},
\cite{Munk&Wunsch1998}, \cite{Burchard&Petersen1999},
\cite{McPhee1999}, \cite{Noh&Kim1999}, \cite{Terray&al.2000},
\cite{Ly&Garwood2000}, \cite{Burchard&Bolding2001},
\cite{Benilov&Ly2002}, \cite{Burchard2001},
\cite{Umlauf&Burchard2001}, \cite{Burchard2002},
\cite{Wunsch&Ferrari2003},  \cite{Ly&Garwood2002},
\cite{Umlauf&Burchard2003}, \cite{Kantha&Carniel2003},
\cite{Umlauf&Burchard2003b}, \cite{Melsom&Saetra2004},
\cite{Gemmrich&Farmer2004}, \cite{Mellor&Blumberg2004},
\cite{Noh2004}, \cite{Kantha&Clayson2004},
\cite{Jezequel&al.2004}, \cite{Carniel&al.2004},
\cite{Kantha2004}, \cite{Canuto&al.2005}, \cite{Noh&al.2005},
\cite{Cheng&al.2005}, \cite{Acreman&Jeffery2007}

\item Langmuir circulations
\cite{Langmuir1938}, \cite{Craik1970}, \cite{Craik1971},
\cite{Garrett1976}, \cite{Craik&Leibovich1976},
\cite{Craik&Leibovich1977}, \cite{Leibovich&Radhakrishnan1977},
\cite{Leibovich1977}, \cite{Faller1978}, \cite{Faller&Caponi1978},
\cite{Weller1978}, \cite{Smith1980}, \cite{Leibovich1980},
\cite{Leibovich1983}, \cite{Faller&Cartwright1983},
\cite{Weber1985}, \cite{Smith&al.1987}, ,
\cite{Nepf&Monismth1991}, \cite{Thorpe1992},
\cite{Skillingstad&Denbo1995}, \cite{Li&Garrett1995},
\cite{Li&al.1995}, \cite{Nepf&al.1995}, \cite{Pluedemann&al.1996},
\cite{Dingemans&al.1996}, \cite{McWilliams&al.1997},
\cite{Graham&Hall1997}, \cite{Smith1998}, \cite{Melville&al.1998},
\cite{Phillips1998}, \cite{Smith1999}, \cite{Araujo&al.2001},
\cite{McWilliams&Sullivan2001}, \cite{Smith2001}
\cite{Veron&Melville2001}, \cite{Tsai2001},
\cite{Weller&Plueddemann2002}, \cite{Groeneweg&Battjes2003},
\cite{Thorpe&al.2003a}, \cite{Thorpe&al.2003b},
\cite{Noh&al.2004}, \cite{Sullivan&al.2004}, \cite{Min&Noh2004},
\cite{Marmorino&al.2005}, \cite{Chini&Leibovich2005},
\cite{Phillips2005}

\item Wave-turbulence interactions and turbulence statistics
\cite{Bowden1950}, \cite{Groen&Dorrestein1950}, \cite{Phillips1958b},
\cite{Phillips1961}, \cite{Stewart&Grant1962}, \cite{Green&al.1972},
\cite{Lumley&Terray1983}, \cite{Kitaigorodskii&Lumley1983},
\cite{Kitaigorodskii&al.1983}, \cite{Cheung&Street1988},
\cite{Magnaudet&Masbernat1990}, \cite{Olmez&Milgram1992}, \cite{Polnikov1993},
\cite{Belcher&Hunt1993}, \cite{Thais&Magnaudet1995},
\cite{Longuet-Higgins1996}, \cite{Thais&Magnaudet1996}, \cite{Milgram1998},
\cite{Nazarenko&al.1999}, \cite{Teixeira2000}, \cite{Thais&al.2001},
\cite{Brocchini&Peregrine2001}, \cite{Teixeira&Belcher2002},
\cite{Brocchini&Peregrine2002a}, \cite{Brocchini&Peregrine2002b},
\cite{Bal&Chou2002}, \cite{Huang&Mei2003}, \cite{Ardhuin&Jenkins2005},
\cite{Ardhuin&Jenkins2006}, \cite{Kantha2005b}, \cite{Babanin2006}

\item Tracer diffusion
\cite{Joseph&Sendner1962}, \cite{Herterich&Hasselmann1982},
\cite{Faller&Auer1988}, \cite{Middleton&Loder1989}, \cite{Mei&Chian1994},
\cite{Balk2002}

\item surface waves - internal waves interactions
\cite{Thorpe1966}, \cite{Gargett&Hughes1972}, \cite{Olbers&Herterich1979},
\cite{Osborne&Burch1980}, \cite{Dysthe&Das1981}, \cite{Agnon&Mei1988},
\cite{Watson1990}, \cite{Kudryavtsev1994}, \cite{McKee1996b},
\cite{Bhattacharyya&Das1997}, \cite{Bhattacharyya&Das1998},
\cite{Donato&al.1999}, \cite{Stocker&Peregrine1999b},
\cite{Bakhanov&Ostrovsky2002}, \cite{Jamali&Seymour2004}

\item acoustic reverberation and sound generation by breaking waves
\cite{Lu&al.1990}, \cite{Mei&Naciri1991}, \cite{Ding&Farmer1994b},
\cite{Felizardo&Melville1995}, \cite{Means&Heitmeyer2002},
\cite{Ma&al.2005}, \cite{Dahl&al.2007}

\item waves and ocean optics \cite{Wing&al.1993}
\end{itemize}

\section{Interaction of waves with floating objects}
\begin{itemize}
\item Damping of waves by viscosity and surface films
\cite{Davies&Vose1965}, \cite{vanDorn1966}, \cite{Alofs&Reisberg1972},
\cite{Lange&Huhnerfuss1984}, \cite{Wu1989b}, \cite{Milgram1998},
\cite{Matar&al.2004}, \cite{Christensen2005}

\item oil dispersion and drift
\cite{Delvigne&Sweeney1988}, \cite{Elliott1991}, \cite{Giarusso&al.2001},
\cite{GarciaSoto2004}

\item Waves and sea ice or ice caps \cite{Greenhill1887}, \cite{Wadhams1978}, \cite{Chu1987}, \cite{Crocker&Wadhams1988}, \cite{Liu&MolloChristensen1988},
\cite{Masson&LeBlond1989}, \cite{Masson1991},
\cite{Bukatov&Bukatova1993}, \cite{Squire&al.1995},
\cite{Meylan&Squire1996}, \cite{Meylan&al.1997},
\cite{Meylan2002}, \cite{Shulz-Stellenfleth&Lehner2002},
\cite{Xia&Shen2002}, \cite{Linton&Chung2003}, \cite{Lavrenov2003},
\cite{Meylan&Masson2006}, \cite{MacAyeal&al.2006},
\cite{Peter&Meylan2007}

\end{itemize}

\section{Interaction of waves with submerged objects}
\cite{Cooker&al.1990}, \cite{Arena2006}

\section{Interaction of waves with surface-piercing obstacles}
\begin{itemize}
\item  engineering \cite{Havelock1929}, \cite{Penney&Price1952},
\cite{Smith&Sprinks1975}, \cite{Yue&Mei1980}, \cite{Mei1985b},
\cite{Agnon&Mei1988}, \cite{Chakrabarti&al.1997},
\cite{McKee1999}, \cite{Longuet-Higgins&Drazen2002}
\end{itemize}
\section{Interaction of waves with the bottom}
\begin{itemize}
\item  steady bottom boundary layer \cite{Hurther&Lemmin2000}, \cite{Nash&Moum2001}

\item  wave bottom boundary layer \cite{Nikuradse1933}, \cite{Reichardt1951}, \cite{Collins1963}, \cite{Jonsson1965}, \cite{Jonsson1967},
\cite{Kajiura1968}, \cite{Wooding&al.1972},
\cite{Jonsson&Carlsen1976}, \cite{Smith&McLean1977},
\cite{McClain&al.1977}, \cite{Soulsby1977},
\cite{Grant&Madsen1979}, \cite{Knight&MacDonald1979},
\cite{Jonsson1980}, \cite{Longuet-Higgins1981},
\cite{Tanaka&Shuto1981}, \cite{Grant&Madsen1982},
\cite{Cacchione&Drake1982}, \cite{Hino&al.1983},
\cite{Trowbridge&Madsen1984a}, \cite{Christoffersen&Jonsson1985},
\cite{Grant&Madsen1986}, \cite{Glen&Grant1987},
\cite{Graber&Madsen1988}, \cite{Davies&al.1988},
\cite{Madsen&al.1988}, \cite{Wilcox1988}, \cite{Jensen&al.1989},
\cite{Graber&al.1989}, \cite{Wilson1989}, \cite{Wiberg&Rubin1989},
\cite{Jensen&al.1989}, \cite{Hara&Mei1990}, \cite{Madsen&al.1990},
\cite{Sleath1991}, \cite{Weber1991}, \cite{Weber1991b},
\cite{Green1992}, \cite{Conley&Inman1992},
\cite{Trowbridge&Agrawal1992}, \cite{Gross&al.1992},
\cite{Wilcox1993}, \cite{Madsen1994}, \cite{Li1994},
\cite{Wiberg1995}, \cite{Mathisen&Madsen1996a},
\cite{Mathisen&Madsen1996b}, \cite{Li&al.1997},
\cite{Mathisen&Madsen1999}, \cite{Foster&al.1999},
\cite{McLean&al.1999}, \cite{Rankin&Hires2000},
\cite{LeHir&al.2000}, \cite{Foster&al.2000},
\cite{Styles&Glenn2000}, \cite{Barrantes&Madsen2000},
\cite{GimenezCurto&Corniero2002}, \cite{DohmenJanssen&Hanes2002},
\cite{Mellor2002}, \cite{Smyth&Hay2002}, \cite{Smyth&Hay2003},
\cite{Trowbridge&Elgar2003}, \cite{Zou&Hay2003},
\cite{NimmoSmith&al.2005}, \cite{Kantha2005c},
\cite{Dohmen-Janssen&Hanes2005}, \cite{Sana&al.2006}

\item bottom friction and spectral wave dissipation (over sand or general)
\cite{Miche1953}, \cite{Zhukovets1963},
\cite{Hasselmann&Collins1968}, \cite{JONSWAP},
\cite{vanIeperen1974}, \cite{Kamphuis1978},
\cite{Hsiao&Shemdin1978}, \cite{Shemdin&al.1978},
\cite{Shemdin&al.1980}, \cite{Miles1983}, \cite{Bouws&Komen1983},
\cite{Weber1988}, \cite{Weber1991}, \cite{Weber1991b},
\cite{Tolman1992a}, \cite{Tolman1993}, \cite{Tolman1994},
\cite{Luo&Monbaliu1994}, \cite{Tolman1995},
\cite{Young&Gorman1995}, \cite{Hendrickson1996},
\cite{Herbers&al.2000a}, \cite{Johnson&Kofoed-Hansen2000},
\cite{Ardhuin&al.2001}, \cite{Ardhuin&al.2002b},
\cite{Ardhuin&al.2003a}, \cite{Ardhuin&al.2003b},
\cite{Mirfenderesk&Young2003}, \cite{Zou2004},
\cite{Longuet-Higgins2005}, \cite{Babanin&al.2005},
\cite{Eshleman&al.2006}, \cite{Keen&al.2007}

\item wave-current bottom boundary layers \cite{Grant&Madsen1979}, \cite{Brevik&Aas1980}, \cite{Brevik1980}, \cite{Grant&Madsen1986},
\cite{Glenn&Grant1987}, \cite{Green&Mccave1988},
\cite{Signell&al.1990}, \cite{Soulsby&al.1993},
\cite{Davies&Lawrence1994}, \cite{Xu&Wright1995},
\cite{Green&McCave2001}, \cite{Jones&Davies2001},
\cite{Myrhaug&al.2001}, \cite{Styles&Glenn2000},
\cite{Styles&Glenn2002}, \cite{Lamb&al.2004},
\cite{Kagan&al.2005}, \cite{Aleksandrova&al.2005},
\cite{Cavallaro&al.2005}, \cite{Musumeci&al.2006}

\item wave attenuation within a submerged canopy and interaction with kelp
\cite{Wing&al.1993}, \cite{Hurd&al.1997}, \cite{Stevens&al.2001},
\cite{Gaylord&al.2003}, \cite{Lowe&al.2007}

\item pore water flow \cite{Precht&Huettel2003}, \cite{Precht&al.2004}

\item bottom friction and spectral wave dissipation over mud
\cite{Forristall&al.1990}, \cite{Sheremet&Stone2003},
\cite{Winterwerp2007}

\item sand ripples \cite{Hunt1882}, \cite{Darwin1883}, \cite{Forel1883},
\cite{Forel1894}, \cite{Ayrton1910}, \cite{Bagnold1946},
\cite{Fauquet1954}, \cite{Bagnold1954}, \cite{Dingler1974},
\cite{Komar&Miller1975}, \cite{Clifton1976}, \cite{Nielsen1979},
\cite{Nielsen1981}, \cite{Tanaka&Shuto1981}, \cite{Blondeaux1990},
\cite{Vittorri&Blondeaux1990}, \cite{Southard&al.1990},
\cite{Anderson&Prell1993}, \cite{Willis&al.1993},
\cite{Vincent&al.1993}, \cite{Wright1993}, \cite{Madsen&al.1993},
\cite{Mogridge&al.1994}, \cite{Wiberg&Harris1994},
\cite{Werner&Kocurek1997}, \cite{Gallagher&al.1998},
\cite{Li&Amos1998}, \cite{Andersen&Fredsoe1999},
\cite{Traykovski&al.1999}, \cite{Fries&al.1999},
\cite{Marsh&al1999}, \cite{Li&Amos1999}, \cite{Sherer1999},
\cite{Andersen1999}, \cite{Andersen2001}, \cite{Andersen&al.2001},
\cite{Hansen&al.2001a}, \cite{Hansen&al.2001},
\cite{Faraci&Foti2001}, \cite{Roos&Blondeaux2001},
\cite{Ardhuin&al.2002}, \cite{Styles&Glenn2002},
\cite{Huang&Dong2002}, \cite{Doucette2002a}, \cite{Doucette2002b},
\cite{GimenezCurto&Corniero2003}, \cite{Crawford&Hay2003},
\cite{Stegner&Wesfreid2004}, \cite{Bundgaard&al.2004},
\cite{Murray&Thieler2004}, \cite{Chang&Hanes2004},
\cite{JarnoDruaux&al.2004}, \cite{Xu2005},
\cite{Sekiguchi&Sunamura2005}, \cite{Williams&al.2005},
\cite{Fernandez&al.2006}

\item wave refraction \cite{Burnside1914}, \cite{Sverdrup&Munk1947}, \cite{Munk&Traylor1947},
\cite{Friedrichs1948b}, \cite{Arthur&al.1952}, \cite{Longuet-Higgins1956}, \cite{Keller1958},
\cite{Dorrestein1960}, \cite{Carrier1966}, \cite{Battjes1968}, \cite{Chao&Pierson1972},
\cite{Krasitskii1974}, \cite{Meyer1979b}, \cite{Kriegsmann1979},
\cite{Miles1985}, \cite{Holthuijsen&Booij1994}, \cite{Ehrenmark1998},
\cite{Peak2004}, \cite{Chen&al.2004}, \cite{Ehrenmark2005}

\item Wave diffraction \cite{Penney&Price1952}, \cite{Rivero&al.1997}, \cite{Booij&al.1997}, \cite{Holthuijsen&al.2003}, \cite{Lavrenov&Park2005}

\item Wave propagation equations (Berkhoff and others) \cite{Berkhoff1972},
\cite{Smith&Sprinks1975}, \cite{Green&Naghdi1976},
\cite{Candel1979}, \cite{Radder1979}, \cite{Lozano&Liu1980},
\cite{Liu&Tsay1983}, \cite{Booij1983}, \cite{Ebersole1985},
\cite{Isobe1986}, \cite{Kirby1986c}, \cite{Kirby1986d},
\cite{Kirby1986e}, \cite{Kirby&Dalrymple1986a},
\cite{Kirby&Dalrymple1986b}, \cite{Dalrymple&Kirby1988},
\cite{Dalrymple&al.1989}, \cite{OReilly1991a},
\cite{OReilly&Guza1991}, \cite{OReilly&Guza1993},
\cite{Massel1993}, \cite{Miles&Zou1993},
\cite{Chamberlain&Porter1995}, \cite{Porter&Staziker1995},
\cite{Kaihatu&Kirby1995}, \cite{Chandrasekera&Cheung1997},
\cite{Suh&al.1997}, \cite{Beji&Nadaoka1997}, \cite{Lee&al.1998},
\cite{Miles&Chamberlain1998}, \cite{Chawla&al.1998},
\cite{Agnon1999}, \cite{Athanassoulis&Belibassakis1999},
\cite{Benoit1999}, \cite{Belibassakis2000}, \cite{Kaihatu2001},
\cite{Eherenmark&Williams2001}, \cite{Belibassakis&al.2001},
\cite{Chandrasekera&Cheung2001}, \cite{Agnon&Pelinovsky2001},
\cite{Shakhin&Shakhina2001}, \cite{Lee&Yoon2004},
\cite{Kim&Bai2004}, \cite{Magne&al.2005a},
\cite{Gerosthathis&al.2005}, \cite{Huang&al.2006}

\item Bottom reflection and scattering \cite{Kreisel1949}, \cite{Anderson1958}, \cite{Takano1960},
\cite{Hasselmann1966}, \cite{Miles1967}, \cite{Mei1969}, \cite{Mei&Black1969},
\cite{Howe1971b}, \cite{Elter&Molyneux1972}, \cite{Long1973},
\cite{Richter1976}, \cite{Fitz-Gerald1976}, \cite{Meyer1979},
\cite{Davies1979}, \cite{Miles1981b}, \cite{Ewing&Pitt1982},
\cite{Heathershaw1982}, \cite{Davies1982}, \cite{King&LeBlond1982},
\cite{Booij1983}, \cite{Kirby&Dalrymple1983}, \cite{Rosales&Papanicolaou1983},
\cite{Liu&Tsay1983}, \cite{Davies&Heathershaw1984}, \cite{Mei&Benmoussa1984},
\cite{Mitra&Greenberg1984}, \cite{Mei1985}, \cite{Miles1985},
\cite{Kirby&Dalrymple1986}, \cite{Kirby1986a}, \cite{Kirby1986b},
\cite{Hara1986}, \cite{Hara&Mei1987}, \cite{Liu1987}, \cite{Yoon&Liu1987},
\cite{Benjamin&al.1987}, \cite{Kirby&al.1987}, \cite{Mei&al.1988},
\cite{Kirby1988}, \cite{Devillard&al.1988}, \cite{Kirby&Vengayil1988},
\cite{Belzons&al.1988}, \cite{Naciri&Mei1988}, \cite{Davies&al.1989},
\cite{Kirby1989}, \cite{Mattioli1991}, \cite{Belzons&al1991}, \cite{Rey1992},
\cite{Rey&al.1992}, \cite{Nachbin&Papanicolaou1992}, \cite{Guazzelli&al.1992},
\cite{Bailard&al1992}, \cite{Miles&Zou1993}, \cite{Kirby1993},
\cite{OHare&Davies1993b}, \cite{Mei&Liu1993}, \cite{Liu&Cho1993},
\cite{Sammarco&al.1994}, \cite{Rey1995}, \cite{Rey&al.1996}, \cite{Maas1996},
\cite{McKee1996b}, \cite{Torres&al.1998}, \cite{Miles&Chamberlain1998},
\cite{Miles1998}, \cite{Liu&Yue1998}, \cite{Pelinovskii&al.1998},
\cite{Chen&Guza1999}, \cite{Yu&Mei2000}, \cite{Agnon&Sheremet2000},
\cite{Ting&al.2000}, \cite{Dulou&al.2000}, \cite{Porter&Porter2000},
\cite{Torres&al.2000}, \cite{Belibassakis2000}, \cite{Porter&Porter2001},
\cite{Stepaniants2001}, \cite{Agnon&Pelinovsky2001}, \cite{Ardhuin2001},
\cite{Ardhuin&Herbers2002}, \cite{Porter2002}, \cite{Pihl&al.2002},
\cite{Belibassakis&Athanassoulis2002}, \cite{Ardhuin&al.2003a},
\cite{Ardhuin&al.2003b}, \cite{Mei&Hancock2003}, \cite{Elgar&al.2003},
\cite{Ye2003}, \cite{Belibassakis&Athanassoulis2004}, \cite{Jeong&al.2004},
\cite{Ruban2004}, \cite{Craig&al.2005}, \cite{Magne&al.2005b},
\cite{Magne&al.2005c}, \cite{Thomson&al.2005}, \cite{Gerosthathis&al.2005},
\cite{Magne2005}, \cite{Wang&al.2006}, \cite{Magne&al.2007},
\cite{Ardhuin&Magne2007}, \cite{Porter&Porter2006}, \cite{Porter&Porter2006b},
\cite{Garnier&Nachbin2006}

\item Wave-induced mean forces on the bottom \cite{Longuet-Higgins1967}, \cite{Mei1973},
\cite{Kachoyan&McKee1985}, \cite{McKee1987b}, \cite{Longuet-Higgins2005}, \cite{Ardhuin2006b}

\item Microseisms \cite{Longuet-Higgins&Ursell1948}, \cite{Longuet-Higgins1950}, \cite{Haubrich&al.1963},
\cite{Hasselmann1963c}, \cite{Grevemeyer&al.2000}

\item Earth's hum and IG waves \cite{Rhie&Romanowicz2006}

\item Bottom topography \textit{per se} and coastline  \cite{Sloss1993}, \cite{Wessel&Smith1996}, \cite{Hall2006}, \cite{Marks&Smith2006}

\item Sandwaves \cite{Nordin&Algert1966}, \cite{Hino1968}, \cite{Dalrymple&al.1978}, \cite{Dyer&Huntley1999}, \cite{Hulscher&vandenBrink2001}, \cite{Idier&al.2002},
\cite{Idier&Astruc2003}, \cite{Fernandez&al.2006},
\cite{Besio&al.2006}

\end{itemize}

\section{Wave statistics outside the surf zone}
\begin{itemize}
\item Wave and crest heights, wave periods \cite{Wiegel1949}, \cite{Longuet-Higgins1952},
\cite{Cartwright&Longuet-Higgins1956},
\cite{Longuet-Higgins1957b}, \cite{Longuet-Higgins1963b},
\cite{Arhan&al.1976}, \cite{Cavanie&al.1976},
\cite{Ezraty&al.1976}, \cite{Arhan&Ezraty1978},
\cite{Battjes1978}, \cite{Longuet-Higgins1980}, \cite{Tayfun1980},
\cite{Larsen1981}, `\cite{Longuet-Higgins1983b},
\cite{Buckley1983}, \cite{Hatori1984}, \cite{Buckley&al.1984},
\cite{Holthuijsen&Herbers1986}, \cite{Myrhaug&Kjeldsen1987},
\cite{Srokosz1998}, \cite{Aage&al.1999}, \cite{Forristall2000},
\cite{Memos&Tzanis2000}, \cite{Ahn2000}, \cite{Prevosto&al.2000},
\cite{Wist&al.2002}, \cite{Memos&al.2002}, \cite{Song&al.2002},
\cite{Fedele&Arena2005}, \cite{Arena2005},
\cite{Soquet-Juglard&al.2005}, \cite{Xu&al.2005}
\item Wave groups \cite{Longuet-Higgins1984b}, \cite{Battjes&vanVledder1984}, \cite{Masson&Chandler1993}, \cite{Cho&Kim2005}
\item Crest length  \cite{Longuet-Higgins1956}
\item Freak waves \cite{Atkins1977}, \cite{White&Fornberg1998}, \cite{Brown2001},
\cite{Onorato&al.2002}, \cite{Prevosto&Bouffandeau2002}, \cite{Janssen2003},
\cite{Onorato&al.2004}, \cite{Fedele&Arena2005}, \cite{Onorato&al.2005},
\cite{Jensen2005}, \cite{Onorato&al.2006}, \cite{Shukla&al.2006}
\item Statistics of other wave properties \cite{Tung&Huang1984}, \cite{Cieslikiewicz&Gudmestad1993}, \textbf{\cite{Phillips&al.1993a}},
\cite{Phillips&al.1993a}
\item long-term statistics \cite{Battjes1984}, \cite{Aage&al.1999}
\end{itemize}

\section{Wave spectral shape and evolution}
\begin{itemize}
\item Fully developed waves \cite{Chase&al.1957}, \cite{Cote&al.1960},
\cite{Kitaigorodskii1962}, \cite{Kitaigorodskii&Strekalov1962},
\cite{Pierson&Moskowitz1964},\cite{Heron1987}, \cite{Ewing&Laing1987},
\cite{Resio&al.1999}, \cite{Alves&al.2003}, \cite{Gourrion2003}

\item Observations of wind-wave growth with fetch \cite{Sverdrup&Munk1947}, \cite{Darbyshire1952}, \cite{Darbyshire1959a},
\cite{Burling1959}, \cite{Darbyshire1959b}, \cite{Hidy&Plate1966},
\cite{Barnett&Wilkerson1967}, \cite{Barnett&Sutherland1968},
\cite{Kononkova&al.1970}, \cite{Liu1970}, \cite{Liu1971},
\cite{Hasselmann1972}, \cite{JONSWAP}, \cite{Toba1973b},
\cite{Toba1974}, \cite{Mitsuyasu&al.1975}, \cite{Seymour1977},
\cite{Kawai&al.1977}, \cite{Crombie&al.1978},
\cite{Stewart&Teague1980}, \cite{Liu&Ross1980}, \cite{Kahma1981},
\cite{Huang&al.1981}, \cite{Cardone&al.1981},
\cite{Holthuijsen1983}, \cite{Hatori1984b},
\cite{Donelan&al.1985}, \cite{Bouws&al.1985},
\cite{Bouws&al.1987}, \cite{Janssen&al.1987},
\cite{Battjes&al.1987}, \cite{Kahma&Donelan1988},
\cite{Walsh&al.1989}, \cite{Dobson&al.1989},
\cite{Anderson&al.1989}, \cite{Perrie&Toulany1990},
\cite{Lewis&Allos1990}, \cite{Ebuchi&al.1992},
\cite{Donelan&al.1992}, \cite{Kahma&Calkoen1992},
\cite{Long&al.1994}, \cite{Kahma&Pettersson1996},
\cite{Young&Verhagen1996a}, \cite{Young&Verhagen1996b},
\cite{Young&al.1996}, \cite{Young1997}, \cite{Young1998},
\cite{Babanin&Soloviev1998}, \cite{Forristall&al.1998},
\cite{Mitsuyasu&Yoshida1998}, \cite{Babanin&Soloviev1998b},
\cite{Watts2003}, \cite{Pettersson2004},
\cite{Pettersson&al.2004}, \cite{Young&Babanin2006a},
\cite{Ardhuin&al.2007}

\item slanting fetch \cite{Donelan&al.1992}, \cite{Walsh&al.1989}, \cite{Pettersson2004},
\cite{Bottema&vanVledder2005}, \cite{Ardhuin&al.2007}


\item Swell evolution \cite{Barber&Ursell1948}, \cite{Munk&Snodgrass1957}, \cite{Munk&al.1963}, \cite{Snodgrass&al.1966},
\cite{Fett&Rabe1976}, \cite{Gjevik&al.1988}, \cite{Stewart&Teague1980},
\cite{Davidan&Lavrenov1991}, \cite{Ursell1999}

\item time-limited growth \cite{Wiegel1961}, \cite{Ivanenkov&al.1977}, \cite{Mitsuyasu&Rikiishi1988}, \cite{Toba&al.1988}, \cite{Hwang&Wang2004}

\item waves in turning winds \cite{Gunther&al.1981}, \cite{Allender&al.1983}, \cite{Holthuijsen&al.1987},
\cite{Young&al.1987}, \cite{Masson1990}, \cite{Vledder&Holthuijsen1993},
\cite{Jackson&Jensen1995}, \cite{Perrie&Toulany1995b},
\cite{Perrie&Toulany1997}, \cite{Jackson&Jensen1999}

\item Wind sea in the presence of swell
\cite{Hatori&al.1981}, \cite{Donelan1987}, \cite{Dobson&al.1989},
\cite{Donelan1999}, \cite{Jenkins2002},
\cite{Violante-Carvalho&al.2004}, \cite{Ardhuin&al.2007}

\item Spectral shape: peak and low frequency cut-off \cite{Zaslavskiy1989}

\item Spectral shape: inertial range and saturation \cite{Phillips1958},
\cite{Pierson&Moskowitz1964}, \cite{Longuet-Higgins1969c},
\cite{Toba1973b}, \cite{DeLeonibus&al.1974},
\cite{Kitaigorodskii&al.1975}, \cite{Thornton1977},
\cite{Forristall1981}, \cite{Kitaigorodskii1983},
\cite{Leykin&Rozenberg1984}, \cite{Phillips1985},
\cite{Kitaigorodskii1987}, \cite{Liu1989}, \cite{Hansen&al.1990},
\cite{Shen&Mei1993}, \cite{Young&al.1995},
\cite{Belcher&Vassilicos1997}, \cite{Ewans1998},
\cite{Hwang&al.2000a}, \cite{Hwang&al.2000b},
\cite{Hara&Belcher2002}, \cite{Smith&Vincent2003},
\cite{Hauser&al.2003}, \cite{Long&Resio2004},
\cite{Long&Resio2007}

\item Spectral shape: High frequency / wavenumber tail and surface slopes \cite{Cox&Munk1954}, \cite{Cox1958},
\cite{Barnett&Wilkerson1967}, \cite{Plant&Wright1979},
\cite{West1981}, \cite{Longuet-Higgins1982}, \cite{West1982},
\cite{Huang&al.1984}, \cite{Banner&Fooks1985}, \cite{Glazman1986},
\cite{Donelan&Pierson1987}, \cite{Donelan&Pierson1987},
\cite{Banner&al.1989}, \cite{Banner1990}, \cite{Hara&al.1994},
\cite{Juszko&al.1995}, \cite{Hwang&al.1996},
\cite{Elfouhaily1997}, \cite{Elfouhaily&al.1997},
\cite{Shaw&Churnside1997}, \cite{Hara&al.1998},
\cite{Lemaire&Sobieski1999}, \cite{Kudryavtsev&al.1999},
\cite{Chapron&al2000}, \cite{Caudal2002}, \cite{Uz&al.2003},
\cite{Plant2003}, \cite{Vandemark&al.2004}, \cite{Zapevalov2005}

\item Hurricane winds and waves \cite{Bretschneider1959}, \cite{Ijima&al.1968}, \cite{Bretschneider1972},
\cite{Borgman1973}, \cite{Bea1974}, \cite{Cardone&al.1976},
\cite{Cardone&al.1977}, \cite{Atkinson&Holliday1977},
\cite{Black1979}, \cite{Holland1980}, \cite{Emanuel1995},
\cite{Powell&al.1996}, \cite{Powell&Houston1996},
\cite{Powell&Houston1998}, \cite{Quilfen&al.1998},
\cite{Wright&al.2001}, \cite{Moon&al.2003},
\cite{Tolman&Alves2005}, \cite{Wang&al.2005},
\cite{Fernandes2005}, \cite{Quilfen&al.2006},
\cite{Aberson&al.2006}, \cite{Young2006}, \cite{Jensen&al.2006}

\item Global or basin-scale wave climate \cite{Dacunha&Hogben1985}, \cite{Challenor&al.1991}, \cite{Carter&al.1991},
\cite{Tournadre1993}, \cite{Gulev&Hasse1998}, \cite{Bauer&Staabs1998},
\cite{Battjes&Groenendijk2000}, \cite{Grevemeyer&al.2000}, \cite{Chen&al.2002},
\cite{Gulev&al.2003}, \cite{Caires&Sterl2003}
 \cite{Caires&Sterl2005}

\item regional wave climate
\cite{Battjes1972b}, \cite{Schopp1983},
\cite{Olaniyan&Afiesimama2002}, \cite{Butel&al.2003},
\cite{Barstow&al.2003}

\end{itemize}

\section{Nearshore hydrodynamics and morphodinamics}
\begin{itemize}
\item Nearshore waves and breaking \cite{Miche1944c}, \cite{Miche1944d},
\cite{Divoky&al.1970}, \cite{Battjes1974b},
\cite{Longuet-Higgins&Fenton1974}, \cite{Witting1975},
\cite{Battjes&Janssen1978}, \cite{Stiassnie&Peregrine1980}

\cite{Thornton&Guza1982}, \cite{Thornton&Guza1983},
\cite{Dally&al.1984}, \cite{Svendsen1984},
\cite{Battjes&Stive1985}, \cite{Stive&Battjes1986},
\cite{Dally&al.1985}, \cite{Svendsen1987},
\cite{Doering&Bowen1987}, \cite{Dally1990}, \cite{Hardy1993},
\cite{Hamm&al.1993}, \cite{Beji&Battjes1994}, \cite{Chen&al.1997},
\cite{Lin&Liu1998}, \cite{Baldock&al.1998},
\cite{Herbers&al.1999}, \cite{Herbers&al.2000},
\cite{Christensen&Deigaard2001}, \cite{Bonneton&al.2001},
\cite{Govender&al.2002}, \cite{Ruessink&al.2003},
\cite{Denny&al.2003},  \cite{Herbers&al.2003},
\cite{Zhao&al.2004}, \cite{Song&Banner2004},
\cite{Biausser&al.2004}, \cite{Stansby&Feng2005}

\item wave statistics \cite{Guza&Thornton1985}, \cite{Doering&Bowen1987}, \cite{Cai&al.1992}, \cite{Baldock&al.1998},
hb\cite{Buccino&Calabrese2002}
\item Wave reflection from shore \cite{Miche1951}, \cite{Carrier&Greenspan1958}, \cite{Snodgrass&al.1962},
\cite{Munk&al.1963}, \cite{Meyer1979b},
\cite{Mahony&Pritchard1980}, \cite{Walton1992},
\cite{Elgar&al.1994}, \cite{OReilly&al.1999}

\item nearshore turbulence \cite{Thornton1979}

\item Infra-gravity waves \cite{Munk1949}, \cite{Tucker1950}, \cite{Biesel1952}, \cite{Longuet-Higgins&Stewart1962},
\cite{Madsen1971}, \cite{Fujinawa1979}, \cite{Huntley&al.1981},
\cite{Mei&Benmoussa1984}, \cite{Yefimov&Soloviev1984},
\cite{Symonds&Bowen1984}, \cite{Holman&Bowen1984},
\cite{HoltmanShay&Guza1987}, \cite{Webb&al.1991},
\cite{Holland&Holman1993}, \cite{Schaffer1993},
\cite{Schaffer1994}, \cite{Herbers&al.1994a}, \cite{Ruessink1998},
\cite{Sheremet&Guza1999}, \cite{Lippmann&al.1999},
\cite{Cadene2000}, \cite{Henderson&al.2001}, \cite{Henderson2002},
\cite{Henderson&Bowen2002}, \cite{Henderson&Bowen2003},
\cite{Aagaard&Bryan2003}, \cite{vanDongeren&al.2003},
\cite{Janssen&Onorato2005}, \cite{vanDongeren&al.2007},
\cite{Karunarathna&CHadwick2007}

\item Run-up \cite{Carrier&Greenspan1958}, \cite{Carrier1966},
\cite{Spielvogel1975} ,\cite{Holland&al.1995}, \cite{Hughes2004},
\cite{Erikson2005}, \cite{Stockdon&al.2006}

\item Wave set-down and set-up \cite{Tancreto1958}, \cite{Saville1961},
\cite{Dorrestein1961}, \cite{Longuet-Higgins&Stewart1963},
\cite{Longuet-Higgins1967}, \cite{Bowen&al.1968},
\cite{Battjes1972c}, \cite{Battjes1974}, \cite{Mei1973},
\cite{Longuet-Higgins1977}, \cite{Guza&Thornton1981},
\cite{Longuet-Higgins1983c}, \cite{Svendsen1984},
\cite{Dolata&Ronsenthal1984}, \cite{Bertotti&Cavaleri1985},
\cite{Holman&Sallenger1985}, \cite{Pechon&al.1997},
\cite{Jonsson1998}, \cite{Lentz&Raubenheimer1999},
\cite{Massel&Gourlay2000}, \cite{Raubenheimer&al.2001},
\cite{Houari&al.2003}, \cite{Cheung&al.2003},
\cite{Feddersen&al.2004}, \cite{Kenyon2004},
\cite{Athanassoulis&al.2004}, \cite{Longuet-Higgins2005},
\cite{Ardhuin2006b}, \cite{Apotsos&al.2007}

\item Nearshore circulation (general) \cite{Bowen1969}, \textbf{\cite{Longuet-Higgins1970}}, \cite{Longuet-Higgins1970b},
\cite{Guza&Thornton1985}, \cite{Thornton&Guza1986},
\cite{Stive&Battjes1986}, \cite{Stive&Wind1986},
\cite{Dingemans&al.1987}, \cite{DeVriend&Stive1987},
\cite{Battjes1988}, \textbf{\cite{Bowen&Holman1989}},
\textbf{\cite{Oltman-Shay&al.1989}}, \cite{Roelvink&Stive1989},
\cite{Dodd&Thornton1990}, \cite{Deigaard&al.1991},
\cite{Church&Thornton1993}, \cite{Whitford&Thornton1993},
\cite{Svendsen&Putrevu1994}, \cite{Haines&Sallenger1994},
\cite{Allen&al.1996}, \cite{Srinivas&Dean1996},
\cite{Reniers&Battjes1997}, \cite{Reniers&al.1997},
\cite{Pechon&al.1997}, \cite{Peregrine1998},
\cite{Li&Dalrymple1998}, \cite{Putrevu&Svendsen1999},
\cite{Falques&al.1999}, \cite{OzkanHaller&Kirby1999},
\cite{Haller&al.1999}, \cite{Feddersen&al.2000},
\cite{GarcezFaria&al.2000}, \cite{Slinn&al.2000},
\cite{Richardson2000}, \cite{Thorpe&Centurioni2000},
\cite{Walstra&al.2001}, \cite{Smith2002}, \cite{Fredsoe&al.2003},
\cite{Chen&al.2003}, \cite{Feddersen&al.2003},
\cite{Feddersen&al.2004}, \cite{Reniers&al.2004a},
\cite{Feddersen&Trowbridge2005}, \cite{Terrile&al.2006},
\cite{Monismith2007}, \cite{Newberger&Allen2007b},
\cite{Reniers&al.2007}, \cite{Feddersen2007}, \cite{Henderson2007}

\item vertical velocity profiles and undertow
\cite{Ozanne&al.2000}

\item Rip currents \cite{Ismail&Wiegel1983}, \cite{Chen&al.1999}, \cite{Haas&Svendsen2000}, \cite{Haller&Dalrymple2001},
\cite{Haas&al.2003}, \cite{MacMahan&al.2003},
\cite{MacMahan&al.2006} ,\cite{Castelle&Bonneton2006}

\item Surf zone macro-vortices \cite{Buhler&Jacobson2001}, \cite{Brocchini&al.2004}, \cite{Kennedy&al.2006}, \cite{Piatella&al.2006}

\item Swash \cite{Ho&Meyer1962}, \cite{Holland&Holman1993},
\cite{Raubenheimer2002},  \cite{Raubenheimer&al.2004}

\item Bio-mecanics of nearshore benthic organisms
\cite{Denny1983}, \cite{Denny1985}, \cite{Denny&Gaines1990},
\cite{Helmuth&Denny2003}, \cite{Harley&Helmuth2003}

\item Nearshore sediment transport \cite{Hsu&Raubenheimer2006}

\item Nearshore morphodynamics \cite{Aubrey&al.1980}, \cite{Thornton&al.1996}, \cite{Srinivas&Dean1996},
\cite{Voulgaris&al.1998}, \cite{Coco&al.2000},
\cite{Falques&al.2000}, \cite{Spielmann&al.2001},
\cite{Shand&al.2001}, \cite{Lafon&al.2002},
\cite{Vincent&Hanes2002}, \cite{Henderson&al.2003},
\cite{Coco&al.2003}, \cite{Hoefel&Elgar2003}, \cite{Coco&al.2004},
\cite{Reniers&al.2004b}, \cite{Castelle2004}, \cite{Rihouey2004},
\cite{MarinoTapia&al.2007}

\item Multiple sand bars \cite{Short1975}, \cite{Symonds&Bowen1984},
\cite{OHare&Davies1993}, \cite{Yu&Mei2000b}, \cite{Dulou&al.2000},
\cite{Mei&al.2001}

\item Waves over coral reefs \cite{Hardy&Young1996},
\cite{Massel&Gourlay2000}, \cite{Falter&al.2004},
\cite{Monismith&Fong2004}

\end{itemize}


\section{Wave and nearshore forecasting}
\begin{itemize}

\item Wind forecasting and analysis methods
\cite{Brocchini&al.1995}
\item Wave forecasting methods \cite{Sverdrup&Munk1947}, \cite{Gelci1949},
\cite{HO1951}, \cite{Pierson1952}, \cite{Beydon1952},
\cite{Bretschneider1952}, \cite{Sallard1954},
\cite{Pierson&al.1955}, \cite{Bretschneider1958}, \cite{CERC1977},
\cite{Gerard1981}, \cite{Cavaleri2006}, \cite{Browne&al.2006}

\item Numerical wave forecasting based moments or 1D spectra
\cite{Chen&Wang1983}, \cite{Holthuijsen&al.1989}

\item Numerical wave forecasting based on energy balance (development) \cite{Gelci&al.1957}, \cite{Barnett1968},
\cite{Longuet-Higgins1957}, \cite{Dobson1967}, \cite{Ewing1971},
\cite{Collins1972}, \cite{Isozaki&Uji1973},
\cite{Hasselmann&al.1976}, \cite{Pierson1977},
\cite{Hasselmann&al.1977}, \cite{Gunther&al.1979},
\cite{Cavaleri&MalanotteRizzoli1981}, \cite{Resio1981},
\cite{Bouws&Battjes1982}, \cite{Golding1983},
\cite{Janssen&al.1984}, \cite{Uji1984}, \cite{Allender&al.1985},
\cite{Komen1985}, \cite{Sobey1986}, \cite{Guillaume1987},
\cite{Young1988}, \cite{WAMDI1988}, \cite{Yamaguchi&al.1988},
\cite{Holthuijsen&deBoer1988}, \cite{Tolman1989},
\cite{Cavaleri&al.1989}, \cite{Holthuijsen&al.1991},
\cite{Tolman1991b}, \cite{Yuan&al.1991}, \cite{Snyder&al.1992},
\cite{Tolman1992b}, \cite{Lin&Huang1996}, \cite{Lin&Huang1996b},
\cite{Tolman&Chalikov1996},
 \cite{Benoit&al.1996}, \cite{Ris1997}, \cite{Gomez&Carretero1997},
\cite{Becq1998}, \cite{OReilly&Wittmann1998},
\cite{Booij&al.1999}, \cite{Andrew1999}, \cite{Monbaliu&al.2000},
 \cite{Ardhuin2001},
\cite{Ardhuin&al.2001}, \cite{Liu&al.2002}, \cite{Tolman2002b},
\cite{Rogers2002}, \cite{Lavrenov2003},
\cite{Holthuijsen&al.2003}, \cite{Hsu&al.2004}, \cite{WISE2007},


\item Source term balance
\cite{Komen&al.1984}, \cite{Tolman&Chalikov1996}, \cite{Bender1996}
\cite{Janssen&al.2005}, \cite{Polnikov2005b}

\item Numerics for hyperbolic problems and transport equations
\cite{Donea1984}, \cite{Saad1986}, \cite{Vichnevetsky1987},
\cite{Cockburn&Shu2001}, \cite{Abgrall2001}, \cite{Cockburn2003},
\cite{Abgrall&Mezine2003}, \cite{Skamarock2006}

\item energy balance modelling: numerics
\cite{Booij&Holthuijsen1987}, \cite{Tolman1995b},
\cite{Lavrenov&Onvlee1995}, \cite{Bender1996},
\cite{Fusina&al.1997}, \cite{Bidlot&al.1997b},
\cite{Tolman&Booij1998}, \cite{Hersbach&Janssen1999},
\cite{Rogers&al.1999}, \cite{Hargreaves&Annan2001},
\cite{Wingeart2001}, \cite{Wingeart&al.2001}, \cite{Tolman2002a},
\cite{Rogers&al.2002}, \cite{Tolman2002}, \cite{Tolman2003},
\cite{Lavrenov2003}, \cite{Bidlot&Janssen2003},
\cite{vanderWesthuysen&al.2004}, \cite{Hsu&al.2005},
\cite{Roland&al.2005}, \cite{Ardhuin&Herbers2005},
\cite{Chawla&Tolman2007}

\item enegy balance modelling: parallelization and code performance
\cite{Tolman2002}, \cite{Tolman2003b}

\item Data assimilation in wave models \cite{LeDimet&Talagrand1986}, \cite{Esteva1988}, \cite{Thomas1988}, \cite{Janssen&al.1989},
\cite{deValk&Calkoen1989}, \cite{Francis&Stratton1990}, \cite{Bauer&al.1992},
\cite{DeLasHeras&Janssen1992}, \cite{Mastenbroek&al.1994},
\cite{DeLasHeras&al.1994}, \cite{Foreman&al.1994}, \cite{deValk&al.1994},
\cite{Lionello&al.1995}, \cite{Bauer&al.1996}, \cite{Lionello&al.1992},
\cite{Bender&Glowacki1996}, \cite{Young&Glowacki1996},
\cite{Holthuijsen&al.1997}, \cite{Hasselmann&al.1997}, \cite{Voorrips&al.1997},
\cite{Dunlap&al.1998}, \cite{OReilly&Guza1998}, \cite{Hersbach1998},
\cite{Breivik&al.1998}, \cite{Voorrips&al.1999}, \cite{Voorrips&al.1999b},
\cite{Greenslade2001}, \cite{Greenslade&Young2004}, \cite{Aouf&al.2006},
\cite{Aouf&al.2006b}, \cite{Keen&al.2007}

\item Validation methods \cite{Tolman1998}, \cite{Marzaban2004}

\item Validation of model winds \cite{Cardone&al.1985}, \cite{Cardone&al.1995},
\cite{Tolman1998}, \cite{Tolman1998b}, \cite{Tang&al.1998},
\cite{Cavaleri&Bertotti2004}, \cite{Signell&al.2005}, \cite{Kepert&al.2005}

\item Validation of wave models \cite{Liu&al.1984}, \cite{Guillaume1990}, \cite{Zambresky1991}, \cite{Cavaleri&al.1991},
\cite{Bertotti&Cavaleri1994}, \cite{Janssen&al.1996},
\cite{Cavaleri&Bertotti1997}, \cite{Cavaleri&al.1997},
\cite{Bidlot&al.1997}, \cite{Heimbach&al.1998},
\cite{Hwang&al.1999}, \cite{Fradon&al.2000}, \cite{Ardhuin2001},
\cite{Tolman2002d}, \cite{Tolman2002e}, \cite{Bidlot&al.2002},
\cite{Bonekamp&al.2002}, \cite{Cavaleri&Bertotti2003},
\cite{Cavaleri&Bertotti2004}, \cite{Ardhuin&al.2003b},
\cite{Chu&al.2004}, \cite{Janssen2004}, \cite{Peak2004},
\cite{Rogers&al.2005}, \cite{Ardhuin2006a},
\cite{Ardhuin&LeBoyer2006}, \cite{Bolanos&Sanchez-Arcilla2006},
\cite{Rogers&al.2007}, \cite{MEWAM2007}, \cite{Ardhuin&al.2007},
\cite{Ris&al.1999}, \cite{Rogers2002}, \cite{Janssen2007},
\cite{Bidlot&al.2007b}

\item Case studies in wave modelling \cite{DellOsso&al.1992}, \cite{Greenslade2001b}

\item Operational wave modelling \cite{Zambresky1987}, \cite{Zambresky1989},
\cite{Bidlot&al.1996},  \cite{Bidlot&al.1997}, \cite{Alves&al.2002}, \cite{Bidlot&al.2002},
\cite{Tolman&al.2002}, \cite{Tolman2002d}, \cite{Bidlot&Janssen2003}, \cite{Cavaleri&Bertotti2004},
\cite{Janssen2004}, \cite{Janssen&al.2005}

\item Coupled air-sea-waves modelling \cite{Dolata&Ronsenthal1984}, \cite{Broche&Forget1992},
\cite{Wu&Flather1992}, \cite{Janssen&al.1992}, \cite{Mastenbroek&al.1993},
\cite{Davies&Lawrence1995}, \cite{Janssen&Viterbo1996}, \cite{Zhang&Li1996},
\cite{Zhang&Li1997}, \cite{Lionello&al.1998}, \cite{Powers&Stoelinga2000},
\cite{Desjardins&al.2000}, \cite{Lalbeharry&al.2000}, \cite{Xie&al.2001},
\cite{Osuna2002}, \cite{Perrie&al.2003}, \cite{Lionello&al.2003},
\cite{Janssen&al.2004}, \cite{Osuna&al.2004}, \cite{Moon2005},
\cite{Janssen2004}, \cite{Ardhuin&al.2005a}

\item Nearshore circulation modelling \cite{Shi&al.2007}

\end{itemize}

\section{Other wave modelling applications}
\begin{itemize}
\item suspended sediment concentration
\cite{Luettich&al.1990}
\end{itemize}


\section{Measurement techniques}
\begin{itemize}
\item general wave observations \cite{Aage&al.1999}, \cite{COST714.2005}
\item in-situ wave measurement
\cite{Barber&al.1946}, \cite{Cote&al.1960}, \cite{Longuet-Higgins&al.1963},
\cite{Cartwright&Smith1964}, \cite{Cavaleri&al.1979}, \cite{Cavaleri&al.1981},
\cite{vanderVlugt&al.1981}, \cite{Ezraty&Cavanie1981},
\cite{Ezraty&Canvanie1983}, \cite{Cavaleri1984}, \cite{Bishop&Donelan1987},
\cite{Allender&al.1989}, \cite{OReilly&al.1996}, \cite{Krogstad&al.1999b},
\cite{Cavaleri2000}, \cite{Graber&al.2000}, \cite{Forristall2000},
\cite{Elgar&al.2001}, \cite{Smith2002b}, \cite{Krogstad2002},
\cite{Pettersson&al.2003}, \cite{Joosten2006}

\item Time series and wave data analysis (general) \cite{Rice1944}, \cite{Woodward1952}, \cite{Blackman&Tukey1958}, \cite{Priestley1965},
\cite{Welsh1967}, \cite{Jenkins&Watts1968},
\cite{Blachman&McAlpine1969}, \cite{Benilov&Flyushkin1970},
\cite{Lii&al.1976}, \cite{Donelan&Pierson1983}, \cite{Young1986},
\cite{Mollo-Christensen1987}, \cite{Goda1988}, \cite{IAHR1989},
\cite{Young1995}, \cite{Gerard&al.1997},
\cite{Hamm&Peronnard1997}, \cite{Hristov&al.1998},
\cite{Krogstad&al.1999}, \cite{DiMarco&al.2001},
\cite{Donelan2002}, \cite{Aarnes&Krogstad2003},
\cite{Elfouhaily&al.2003}, \cite{Zhang&Zhang2006}

\item Estimation of the directional wave spectrum from point measurements \cite{Barber1963}, \cite{Davis&Regier1977},
\cite{Long&Hasselmann1979}, \cite{Long1980}, \cite{Jefferys&al1981}, \cite{Pawka1983}, \cite{Isobe&al.1984},
\cite{Lygre&Krogstad1986}, \cite{Marsden&Juszko1987}, \cite{Kuik&al.1988}, \cite{Nwogu1989},
\cite{Herbers&Guza1990}, \cite{Long&Atmadja1994}, \cite{Donelan&al.1996}, \cite{Benoit&al.1997},
\cite{Carvalho&Parente2000}, \cite{Donelan2002}, \cite{Zhang&Zhang2006}

\item Partitionning of directional wave spectra
\cite{Wang&Hwang2001}, \cite{Hanson&Phillips2001}

\item Bispectral analysis \cite{Lii&al.1976}, \cite{Imasato&Kunishi1977}, \cite{Kim&Powers1979}, \cite{Elgar&al.1995}

\item Estimation of wave reflection \cite{Isaacson1991}, \cite{Baldock&Simmonds1999}, \cite{Rey&al.2002}

\item Photogrammetry \cite{Chase&al.1957}, \cite{Cote&al.1960}, \cite{Holthuijsen1983}, \cite{Holthuijsen1983b}

\item Wave spectra from range measurements \cite{Walsh&al.1985}

\item bathymetry and currents from wave remote sensing \cite{Dugan&al.1996}, \cite{Grilli1998}, \cite{Hessner&al.1999}, \cite{Stockdon&Holman2000},
\cite{Holland2001}, \cite{Dugan&al.2001b}, \cite{Misra&al.2003},
\cite{Aarninkhof&al.2003}, \cite{McNinch2007}

\item optical methods \cite{Monaldo&Kasevich1981}, \cite{Dugan&al.1996}, \cite{Dugan&al.2001a},
\cite{Dugan&al.2001b}, \cite{Gelpi&al.2001}, \cite{Terrill&al.2001}

\item HF-VHF-UHF radar (grazing) \cite{Wait1966}, \cite{Tveten1967},
\cite{Crombie&Watts1968}, \cite{Ward1969},
\cite{Munk&Nierenberg1969}, \cite{Peterson&al.1970},
\cite{Hasselmann1971}, \cite{Barrick1972}, \cite{Barrick&al.1974},
\cite{Stewart&Joy1974}, \cite{Barrick1977}, \cite{Forget&al.1981},
\cite{Janopaul&al.1982}, \cite{Broche&al.1983},
\cite{Prandle&Ryder1985}, \cite{Lawrence&Smith1986},
\cite{Schott&al.1986},  \cite{Broche&al.1986},
\cite{Barrick&Lipa1986}, \cite{Prandle1987},
\cite{Broche&Forget1993}, \cite{Fernandez&al.1996},
\cite{Lewis&al.1998}, \cite{deValk&al.1999}, \cite{Wyatt2000},
\cite{Essen&Gurgel2000}, \cite{Shrira&al.2001},
\cite{Breivik&Saetra2001}, \cite{Wyatt2002}, \cite{Wyatt2003},
\cite{Forget&al.2003}, \cite{Green2003}, \cite{Vesecky&al.2004},
\cite{Ivonin&al.2004}, \cite{Emery&al.2004}, \cite{Wyatt&al.2005},
\cite{Green&Wyatt2006}, \cite{Hisaki2006}, \cite{Forget&al.2006},
\cite{Haus2007}, \cite{Wyatt&al.2007}

\item radar backscatter \cite{Rice1951}, \cite{Barrick&Peake1968}, \cite{Barrick1968}, \cite{Guinard&al.1971},
\cite{Hasselmann1972}, \cite{Jackson1979}, \cite{Plant&Keller1983},
\cite{Keller&al.1985}, \cite{Banner&Fooks1985}, \cite{Holliday1987},
 \cite{Donelan&Pierson1987}, \cite{Phillips1988}, \cite{Winebrenner&Hasselmann1988}, \cite{Jeynes1989}, \cite{Wu1990},
 \cite{Jessup&al.1990}, \cite{Jessup&al.1991a}, \cite{Jessup&al.1991b}, \cite{Plant&al.1994},
 \cite{Thompson&Gotwols1994}, \cite{Gotwols&Thompson1994}, \cite{Apel1994}, \cite{Colton&al.1995},
\cite{Barrick1995}, \cite{Hauser&Caudal1996}, \cite{Voronovich1996},
\cite{Ericson&al.1999}, \cite{Quilfen&al.1999},  \cite{Plant&al.1999},
 \cite{Phillips&al.2001}, \cite{McDaniel2001}, \cite{Bourlier&al.2002}, \cite{Fuks&Voronovich2002}, \cite{Plant2002},
\cite{Voronovitch2002}, \cite{Elfouhaily&al.2002},
\cite{Guerin2002}, \cite{Bourlier&Berginc2003}, \cite{Plant2003},
\cite{Kudryavtsev&al.2003a}, \cite{Elfouhaily&al.2003a},
\cite{Elfouhaily&al.2003b}, \cite{Elfouhaily&al.2003c},
\cite{Elfouhaily&al.2003d}, \cite{Elfouhaily&Guerin2004},
\cite{Guerin&al.2004}, \cite{Elfouhaily&al.2004},
\cite{Mouche&al.2005}, \cite{Mouche&al.2006},
\cite{Gardashov2006}, \cite{Mouche&al.2007}

\item radar modulations \cite{Plant&al.1978}, \cite{Wright&al.1980}, \cite{Smith1986}, \cite{Feindt&al.1986},
\cite{Keller&al.1994}, \cite{Romeiser&al.1994}, \cite{Kudryavtsev&al.2003b}

\item RAR \cite{Jackson&al.1985}, \cite{Jackson&Walton1985}, \cite{Hauser&al.1992}, \cite{Askari&Keller1994},
\cite{Long&al.1994},
 \cite{Pettersson&al.2003}, \cite{Aouf&al.2006b}

\item Altimetry for wind and waves
\cite{Queffeulou1980}, \cite{Queffeulou1983}, \cite{Monaldo1988},
\cite{Tournadre&Ezraty1990}, \cite{Ebuchi&al.1992},
\cite{Tournadre&Blanquet1994}, \cite{Callahan&al.1994},
\cite{Cotton&Carter1994}, \cite{Dobson&Monaldo1996},
\cite{Janssen&al.1997}, \cite{Challenor&Cotton1997},
\cite{Krogstad&Barstow1999}, \cite{Quartly2000},
\cite{NAVONOAA2002}, \cite{NAVONOAA2002}, \cite{Gourrion2003},
\cite{Picot&al.2003}, \cite{Gommenginger&al.2003},
\cite{Queffeulou2004}, \cite{Queffeulou2006},
\cite{Tolman&al.2007}

\item Altimetry sea-state bias
\cite{Glazman&Srokosz1991}, \cite{Gaspar&al.1994}, \cite{Elfouhaily&al.1999},
\cite{Gaspar&al.1994}, \cite{Gaspar&Florens1998}, \cite{Elfouhaily&al.2001},
\cite{Chapron&al.2001a}, \cite{Vandemark&al.2002}, \cite{Chambers&al.2003},
\cite{Feng&al.2004}, \cite{Vandemark&al.2005}, \cite{Feng&al.2006}

\item Scatterometry \cite{Ezraty1986}, \cite{Halpern&al.1983}, \cite{Quilfen&al.1998}, \cite{Quilfen&al.2001},
\cite{Chelton&al.2004}, \cite{Hashizume&Liu2004},
\cite{Suzuki&al.2007}, \cite{Quilfen&al.2007}

\item Radiometry for winds \cite{Yueh&al.2006}, \cite{Bettenhausen&al.2006}, \cite{Brown&al.2006}, \cite{Monaldo.2006},
\cite{Freilich&Vanhoff2006}, \cite{Turk&al.2006}, \cite{Adams&al.2006}, \cite{English&al.2006},

\item SAR speckle over the ocean \cite{Ouchi&Wang2005}, \cite{Kanevsky2005}

\item SAR and ocean features \cite{Strong&DeRycke1973}, \cite{Vesecky&Stewart1982},
\cite{Fu&Holt1983}, \cite{Thompson&Gasparovic1986}, \cite{Munk&al.2000},
\cite{Larson&al.1976}, \cite{Alpers1985}, \cite{Johannessen&al.1991},
\cite{Lyzenga1991}, \cite{Mognard&al.1991}, \cite{Gower1993},
\cite{Johannessen&al.1996}, \cite{Marmorino&al.1997},
\cite{Ufermann&Romeiser1999}, \cite{Ufermann&Romeiser1999b},
\cite{Porter&al.2001}, \cite{Lin&Chubb2003}

\item SAR for wind and waves \cite{Elachi&al.1977}, \cite{Alpers&al.1981}, \cite{Meadows&al.1982},
\cite{Alpers&Hasselmann1982}, \cite{Alpers1983},
\cite{McLeish&Ross1983}, \cite{Hasselmann&al.1985},
\cite{Beal&al.1986}, \cite{Holt&Gonzalez1986},
\cite{Alpers&Bruning1986}, \cite{Holt1988},
\cite{Irvine&Tilley1988}, \cite{Bruning&al.1990},
\cite{Hasselmann&Hasselman1991}, \cite{Marom&al.1990},
\cite{Marom&al.1991}, \cite{Krogstad1992},
\cite{Engen&Johnsen1995}, \cite{Forget&al.1995},
\cite{Kerbaol&al.1998}, \cite{Engen&al.1998},
\cite{Krogstad&Barstow1999}, \cite{Chapron&al.2001b},
\cite{Vandemark&al.2001}, \cite{Voorrips&al.2001},
\cite{Shulz-Stellenfleth&Lehner2002}, \cite{Portabella&al.2002},
\cite{Dankert&al.2003}, \cite{Shulz-Stellenfleth&Lehner2004},
\cite{Borge&al.2004b}, \cite{Ardhuin&al.2004d},
\cite{Collard&al.2005}, \cite{Sikora&al.2006}
\cite{Schulz-Stellenfleth&al.2007}

\item InSAR and microwave Doppler analysis \cite{Wright&Keller1971},
\cite{Schuler1978}, \cite{Plant&Wright1979}$^\star$, \cite{Shuchman1979},
\cite{Plant&Schuler1980}, \cite{Plant&Wright1980}$^\star$,
\cite{Keller&al.1986}, \cite{Goldstein&Zebker1987}, \cite{Thompson1989},
\cite{Madsen1989}, \cite{Plant&Keller1990}, \cite{Marom&al.1990},
\cite{Thompson&al.1991}, \cite{Marom&al.1991}, \cite{Bamler1991},
\cite{Shemer&al.1993}, \cite{Thompson&Jensen1993}, \cite{Chapman&al.1994},
\cite{Gatelli&al.1994}, \cite{Ainsworth&al.1995}, \cite{Smith&al.1996},
\cite{Graber&al.1996}, \cite{Moller&al.1998}, \cite{Romeiser&Thompson2000},
\cite{Frasier&Camps2001}, \cite{vanderKooij&al.2001}, \cite{Romeiser&al.2003a},
\cite{Romeiser&al.2003b}, \cite{Chapron&al.2004}, \cite{Lombardini&al.2004},
\cite{Plant&al.2005}, \cite{Chapron&al.2005}, \cite{Romeiser&al.2005}, \cite{Farquharson&al.2005}

\item X-band radar \cite{Kudryavtsev&al.1995}, \cite{Borge&al.1999}, \cite{Gangeskar2000}, \cite{Wolf&Bell2001},
\cite{Senet&al.2001}, \cite{Haller&Lyzenga2003}, \cite{Dankert&Rosenthal2004},
\cite{Borge&al.2004}, \cite{Dankert&Rosenthal2004b}, \cite{Farquharson&al.2005}

\item GNSS reflections \cite{Lowe&al.2002}, \cite{Rius&al.2002}

\item Atmospheric infrasound \cite{Garces&al.2003}
\end{itemize}

%%%\bibliographystyle{apalike}

\bibliographystyle{alpha_url}
%\bibliographystyle{ieeetr_url}
\bibliography{wave}


\end{document}
